Do Robots Increase Wealth Dispersion?*

THOMAS JANSSON and YIGITCAN KARABULUT

This Version: April 29, 2019

First Draft: August 13, 2018

Abstract

We demonstrate that increased automation has a significant impact on both static and dynamic aspects of wealth distribution. Households who are more exposed to robots at work accumulate less wealth and experience greater downward mobility in the wealth distribution. The negative wealth effects of robots are not merely a consequence of differences in earned incomes or in saving rates. We argue and provide evidence that the adverse effects of rapid robotization on individual workers' human capital, and thereby, on their financial risk taking behavior and investment choices appear to be an additional operative channel.

Keywords: Wealth inequality, portfolio choice, financial wealth, robots, automation, labor income uncertainty.

JEL Codes: D31, J24, E21, D1, G11

^{*}Thomas Jansson: Sveriges Riksbank (thomas.jansson@riksbank.se); and Yigitcan Karabulut: Frankfurt School of Finance and Management, and CEPR (y.karabulut@fs.de). We would like to thank Pooyan Amir Ahmadi, Greg Howard, Claduia Malacuso, Anna Salomons, Pascual Restrepo, Francisco Urzua-Infante, Chen Yeh, Ulrich Zierahn, and participants at the Tasks Conference on "Robotics, Artificial Intelligence and the Future of Work" in Bonn, International Conference on "Technology, Demograhics, and the Labor Market" in Cologne, and seminar participants at the University of Illinois at Urbana-Champaign for their valuable comments and suggestions. The views described in this paper are the authors' own and do not necessarily reflect those of the Sveriges Riksbank.

1 Introduction

In recent years we have witnessed an accelerated progress in digital technologies and significant advances in robotics and other related technologies. According to International Federation of Robotics (IFR), the worldwide industrial robot stock has almost tripled in the past decade, and is projected to grow at a similar (or even a faster) rate over the next ten years.¹ The extent and rapidity of the progress in automation, including the major leaps in artificial intelligence capabilities, raises several questions with important implications for individuals.² What are the consequences of rapid adoption of robots for wages and employment prospects of individual workers? Does its impact extend beyond the labor market to other dimensions of economic and financial behavior of individuals? Who are the winners and losers of increased automation?

In this paper we empirically explore these questions with a particular focus on the effects of automation on household wealth accumulation, and on potential mechanisms through which pervasive automation contributes to changes in the distribution of wealth, and the implications of these effects for the evolution of wealth inequality.

We operationalize increased importance of automation by focusing on an industry-level measure of robot use. Specifically, we consider adoption of industrial robots, which are defined as reprogrammable and fully autonomous machines that are capable of being adapted to perform different tasks (Acemoglu and Restrepo, 2017; Graetz and Michaels, 2017; IFR, 2017). We combine this industry-level measure of automation with an extensive individuallevel panel dataset that contains detailed wealth records and highly accurate information on the demographics and labor market outcomes of approximately 300,000 individuals between 1999 and 2007.

¹See also Acemoglu and Restrepo (2017) and the references therein.

²Indeed, there is a burgeoning literature that focuses on the economic consequences of rapid automation, with a particular emphasis on its effects on the labor market. The recent evidence suggests that, despite their positive impact on productivity (Graetz and Michaels, 2017), automation and advances in production methods negatively affect wages and employment opportunities of individual workers (Acemoglu and Restrepo, 2017; Autor and Salomons, 2018), and further correlate with increases in labor income risk and wage inequality (Kogan et al., 2018).

We provide evidence that increased automation has indeed negative effects on both static and dynamic distribution of wealth. Specifically, increased exposure to robots at work significantly lowers the percentile wealth rank of individuals within their birth cohort-year distributions and further increases the probability of downward mobility in the wealth distribution over the sample period. The magnitude of these effects is also meaningful in economic terms. To give an idea of the magnitude of robotization effects, we find that a one-standard deviation exogenous rise in the robot density between 1999 and 2007, which corresponds to an increase of 3.27 robots per thousand workers in a given industry, reduces the rank of individuals in the wealth distribution by 2.5 percentiles. This effect is present, controlling for a wide range of observable household characteristics, industry-level changes and trends, as well as for region specific macro conditions and shocks.

The implication of the negative wealth effects is mirrored in Figure 1 that provides some striking (though informal) evidence on the relationship between increased use of robots and evolution of wealth inequality. We see that the rise in wealth inequality, as measured by the interquartile range of household net wealth, monotonically increases in the changes in robot density across industries, which suggests that rapid automation is likely to play a role in increased wealth dispersion.

To understand the mechanism underlying these results, we first investigate whether the impact of robots on household wealth operates through its direct effects on earnings of individuals. Indeed, prior literature documents significant negative effects of automation on wages and employment prospects. For example, Acemoglu and Restrepo (2017) show that increased use of industrial robots across US commuting zones contributes to lower aggregate wages and employment.³ Even though we find significant negative wage effects of automation, our analysis reveals that differences in earned incomes alone do not explain the variation in levels and dynamics of household wealth.

 $^{^{3}}$ Acemoglu and Restrepo (2017) quantify the employment effects of automation between 1993 and 2007, and predict that adoption of industrial robots reduces aggregate employment in the US by 414,000 to 756,000 workers during this time period.

Next, we examine the potential role of heterogeneity in saving behavior across households to explain the observed patterns in the wealth distribution. We document that increased automation still exhibits negative significant effects on household wealth even when we control for differences in initial (active and total) saving rates or new saving flows of households. This suggests that heterogeneity in saving behavior is not the only mechanism driving our results.

In fact, our analysis indicates that adverse effects of robotization on individual workers' human capital, and thereby, on their financial risk taking behavior and investment choices appear to be an operative channel. In particular, we first document that rapid adoption of robots leads individuals to face significantly higher background labor income risk, which is measured by job-loss risk of individuals.⁴ We then show that households who are more exposed to robots at their work, and hence, face increased uninsurable risk in their human capital are more likely to exit from the stock market. Specifically, ceteris paribus, a one-standard-deviation exogenous rise in the robot density in their industry of employment leads to an 11.4% increase in the exit probability of households from the stock market. As these individuals fully rebalance their financial portfolio away from (both direct and indirect) stocks, they also forgo substantial equity returns up to 4.3% on a year during the observation period (Calvet et al., 2007), and thereby, experience a substantial drop in the growth of their financial wealth and accumulate less financial wealth relative to their income.⁵

A natural question is whether there are any asymmetries in the distribution of negative wealth effects of automation across individuals. While skill upgrading of jobs as a result of emerging new technologies could favor some people, it can leave behind others, notably,

⁴This result is further strengthened by Kogan et al. (2018) who find that automation, or more generally, advances in production methods are associated with substantial increases in the labor income risk of individual workers.

⁵These findings also relate to the literature that focuses on the mechanisms that can generate observed patterns in the distribution of wealth. A recent strand of this literature emphasizes heterogeneity in returns to wealth as an important driver of wealth inequality (Benhabib et al., 2011; Gabaix et al., 2016; Bach et al., 2016; Fagereng et al., 2018). For example, Benhabib et al. (2017) examine the contributions of different factors to top wealth concentration, and show that skewed and persistent distribution of earnings, differences in saving rates as well as heterogeneous returns to capital are all necessary and important to account for high concentration of wealth.

those individuals with lower human capital or skills (Brynjolfsson and McAfee, 2012; Autor, 2015; Sachs et al., 2015; Berg et al., 2018). For example, Acemoglu and Restrepo (2018b) show that middle-aged workers who perform blue-collar tasks are more likely to be replaced by robots relative to older workers with specialized knowledge in non-production services. Similarly, according to the estimates of the US Council of Economic Advisers, more than 80 percent of jobs paying less than 20 USD per hour in the US would be negatively affected by automation, whereas this number accounts for 4 percent of jobs making above 40 USD per hour.⁶ Hence, it is conceivable to argue that automation and advances in production methods can have asymmetric effects on the economic and financial behavior of individuals, depending on the required skill-level or type of their occupation.

When we study the distributional effects of automation, we only observe negative significant wealth effects for low-skill individuals (i.e., proxied by lower-educational attainment), while we find no effect for high-skills individuals. Similarly, the reduced financial risk taking channel is only operative for the low-skill individuals, presumably because robots are more likely to render the skills of low-educated workers obsolete, and hence, these individuals face a higher idiosyncratic labor income risk. Overall, our findings suggest that rapid automation can widen, the already large and persistent, wealth gap between high- and lowskill individuals, and further caution against the potential distributional challenges created by increased use of robots.

In our empirical analysis, the panel dimension and detailed nature of our dataset allow us to explicitly account for a wide range of relevant individual characteristics and further to isolate the effects of robots from various other industry-level changes and region specific macro shocks. Still, a threat to our identification is posed by the presence of unobserved industry shocks. To overcome this endogeneity issue, following a similar identification strategy as in Autor et al. (2013) and Acemoglu and Restrepo (2017), we instrument for changes

⁶See the report on "Artificial Intelligence, Automation, the Economy" by White and House's Economic Advisers, Council of December 2016. The report \mathbf{is} available at https://obamawhitehouse.archives.gov/blog/2016/12/20/artificial-intelligence-automation-and-economy.

in robot density in the Swedish industries, that is the country of interest in our analysis, using the contemporaneous median changes in robot density across eleven other Western European countries. Our identification strategy proceeds from the notion that the adoption of robots in these eleven developed countries represents the advances in global technological frontier (Acemoglu and Restrepo, 2017). Hence, our instrumental variable strategy enables us to identify the exogenous variation in the adoption of robots in the Swedish industries induced by improvements in the technological frontier of robotics in the corresponding industries and to pin down its causal effects on household wealth accumulation.

Our paper complements a small but growing literature which analyzes the economic consequences of increased automation.⁷ For example, Acemoglu and Restrepo (2017) find that penetration of industrial robots across US local labor markets reduce aggregate employment and wages, while, in an international sample, Graetz and Michaels (2017) document positive productivity effects of automation, which, however, reduce the employment of low-skill workers. We contribute to this literature along several dimensions. First, our micro-data evidence on the negative wage and employment effects parallels and complements the findings of these studies in a different country and time period. Second, and more importantly, we provide the first direct evidence that negative effects of robotization extend beyond the labor market to the dynamics of wealth accumulation, explore the potential underlying mechanisms, and further show its implications for wealth inequality.⁸

Our work also links to the empirical literature on the importance of uninsurable background risk in the demand for risky assets (Fagereng et al., 2017; Betermier et al., 2012). Specifically, we identify an additional source of idiosyncratic labor income risk, which is likely to become more important in the future, and document its effects on the portfolio choice and financial wealth accumulation of individuals.

⁷See also Acemoglu and Restrepo (2018a); Aghion et al. (2017); Freeman (2015); Benzell et al. (2015); Sachs and Kotlikoff (2012).

⁸It is, however, important to note that the focus of our paper is on the direct effects of automation on economic and financial outcomes of households who are exposed to robots at work rather than the aggregate impact of robots as in, for example, Acemoglu and Restrepo (2017).

The remainder of the paper proceeds as follows. Section 2 introduces the data sources, and presents information the construction of the main variables of interest in our empirical analysis. In Section 3 we discuss the econometric challenges in the analysis, and explain how we tackle them. Section 4 presents the impact of increased automation on labor market. In Section 5 we present the results on the wealth analysis and further discuss and explore mechanisms. Section 6 reports the findings on the distributional effects of robots. In Section 7 we present the results of additional robustness and sensitivity checks. Section 8 concludes the paper.

2 Data, Variable Definitions, and Descriptive Statistics

In the analysis of the effects of automation on economic behavior and wealth accumulation of households, we make use of various information from several sources. In what follows, we describe our data sources, provide detailed information on our main variables of interest, and present descriptive statistics for the sample.

To measure the degree of automation, we acquire time-series data on the stock of industrial robots disaggregated at the industry level from the International Federation of Robotics (IFR).⁹ The IFR collects annual information on the total stock of robots and new robot installations, detailed at the 2-digit industry level, for approximately 50 countries since 1993 by surveying the robot producers and suppliers around the world (Graetz and Michaels, 2017; Acemoglu and Restrepo, 2017; IFR, 2017).¹⁰ For Sweden, on which we focus our investigation, we observe the total stock of robots for 14 industries on a yearly basis for the time

⁹Industrial robots are defined as reprogrammable and fully autonomous machines that are capable of being adapted to perform different tasks (i.e., being multipurpose) (IFR, 2017). According to the definitions of the IFR (2017), reprogrammable means that "robots are designed so that programmed motions or auxiliary functions can be changed without physical alteration", and multipurpose refers to the "capability of being adapted to a different application with physical alterations".

¹⁰The IFR also provides information on the application of robots (e.g., handling, dispensing, processing), though, there is no industry breakdown for this particular information, and it is only available at the country level. According to the information provided by the IFR, as of 1999, the application areas of industrial robots that we consider in our analysis are as follows: 63% of industrial robots are used in handling and machine tending; 22% are in welding and soldering; 7% are in assembling and disassembling; 2.3% are in dispensing, and 2% are in processing.

period between 1993 and 2016.¹¹ These industries include agriculture, forestry, fishing; mining and quarrying; manufacturing; utilities; construction; and education, research, and development. For the manufacturing industry, we have a more detailed breakdown of industries approximately at the 3-digit level, which includes along others food and beverages; textiles; wood and furniture; basic metal and metal products; electrical and electronics; and automative industries.¹² In Table I, we provide information on the use of industrial robots and number of workers for the Swedish industries during our sample period.

Following Acemoglu and Restrepo (2017), we merge the industry-level robot data with the number of workers in each corresponding industry, which we collect from the *EU KLEMS* dataset (Jäger, 2016). We then compute the robot density per thousand workers for each industry in a given year as follows:

$$Robot_Density_{jt} = \frac{\text{No of Robots}_{jt}}{\text{No of Workers}_{jt}}$$
(1)

As presented in Table I, automative industry has the highest robot density with 27.86 robots per thousand workers, which is followed by basic metal and metal products industry with 11.35 robots per thousand workers as of 1999.

We next merge the processed data on stock of robots with the household-level LINDA dataset, which is provided by Statistics Sweden.¹³ LINDA consists of an annual cross-sectional sample of around 300,000 individuals, or approximately 3% of the entire Swedish population. The data contain highly accurate information on financial (e.g., detailed decomposition of household wealth at the individual-security level) and demographic characteristics (e.g., age, marital status, educational attainment) of each sampled individual as well

¹¹See World Robotics 2017 report from the IFR for the details on the calculation of the inventory of robots.

¹²The IFR also reports the number of robots that are not classified into any industry. To minimize potential misclassification and measurement errors, we do not consider those values that fall into the "Unspecified" category when computing the robot exposure variable.

¹³It is important to note that the IFR and Statistics Sweden use different industry classifications. For example, LINDA dataset provides information for households' industry of occupation as detailed as at the 5-digit level. We follow a similar matching procedure as in Graetz and Michaels (2017). We provide further details about the matching procedure in Table O.A.1 in the Online Appendix.

as very detailed information on their labor market outcomes such as earnings, employment status, and industry of occupation (i.e., detailed at the three-digit SNI code) for the period from 1999 to 2007.

When constructing the working sample, we adopt a conservative strategy in order to minimize potential misclassification or measurement errors. First, we only focus on workingage households who age between 22 and 60 during the sample period (Hurst and Lusardi, 2004; Autor et al., 2013). Second, we exclude from the sample households who are classified as student, housemaker, self-employed, unemployed or retired, focussing only on the employed individuals in the initial period. Next, we restrict our attention only to those households who are employed in industries, which are directly affected by adoption of industrial robots and for which the IFR provides information on the number of robot stock.¹⁴ According to the employment counts from the EU KLEMS database, the number of employees in the industries that we consider in our analysis represents 55.5% of the workers in the market economy and 35% of the workers in all industries in Sweden. When constructing the final sample, we require households to be employed in certain industries only in the initial period, however, we allow for the sampled households to switch industries or become unemployed or move to different locations in the following years. Finally, out of this conservatively constructed sample, we eliminate households with any missing information on labor market outcomes, financial assets and demographics. Overall, our final sample comprises 30,375 households in any given year between 1999 and 2007. Descriptive statistics on the financial and demographic characteristics of the households are presented in Panel A of Table II.¹⁵

The key variable of interest in our analysis is the household's exposure to robots, which

¹⁴In other words, we exclude from the sample households who are employed in industries such as information and communication; community social and personal services; and other service activities. Alternatively, we could have set the value for stock of robots for those industries to zero, and include the households from those industries in the empirical analysis. We opt out for that approach, as we want to minimize the measurement error in household's exposure to robots, which is the key variable of interest in our empirical analysis.

¹⁵It is important to note that, as of now, we only consider the household head's exposure to robots (rather than each household member's exposure to robots) when analyzing the effects of automation on household economic behavior.

we define at the industry level as follows:

$$\Delta Robot_Density_{j}^{99 \to 07} = \frac{\text{No of Robots}_{j}^{07}}{\text{No of Workers}_{j}^{95}} - \frac{\text{No of Robots}_{j}^{99}}{\text{No of Workers}_{j}^{95}}$$
(2)

In our empirical analysis, we focus on the effect of long-differences in exposure to robots on various dimensions of household economic behavior, hence, we consider the changes in robot density in a given industry between 1999 and 2007. Note that we use the number of workers in 1995 (rather than the contemporaneous values) as our baseline employment level when constructing our variable to limit the potential simultaneity bias among employment and adoption of robots, mainly because current employment levels may be affected by the anticipation of increased automation (Acemoglu and Restrepo, 2017).¹⁶ Panel B of Table II provides some detailed information on our key variable. We observe an increase in the number of robots per thousand workers between 1999 and 2007 with a mean (standard deviation) value of 2.69 (3.27) with the rubber, plastic and chemical product industry experiencing the largest growth (i.e., 11.5 robots per thousand workers) during the observation period.

As our main variable of interest is defined at the industry level, we control for numerous industry characteristics in our analysis to isolate the effects of exposure to robots from other industry-wide changes and trends. First, we construct a control variable for exposure to imports from China at the industry level. Previous literature shows that increased imports from China (and other low-wage countries) have a negative effect on the employment, wages, and labor-force participation (Autor et al., 2013; Bloom et al., 2016). To address this issue, following Autor et al. (2013), we construct a variable that captures the changes in the exposure to Chinese imports per thousand workers between 1999 and 2007 as follows:

 $^{^{16}}$ Even though the time-series data on the number of industrial employees in the *EU KLEMS* database are available from 1993 onwards for Sweden, we use 1995 as our base employment year mainly because of the data availability for other developed European countries. In later stages of the paper, we use the exposure to robots from various developed countries as an instrument for that of in Sweden to address the potential endogeneity issue.

$$\Delta Chinese_Import_{j}^{99 \to 07} = \frac{\Delta Imports_{j}^{99 \to 07}}{\text{No of Workers}_{j}^{95}}$$
(3)

where $\Delta Imports_j^{99\to07}$ is the changes in imports from China (measured in Swedish Kroner in thousands) in industry *j* between 1999 and 2007. We normalize this variable by the employment levels (in thousands) in industry *j* from 1995. The import data are collected from Statistics Sweden, and information on employment levels again comes from the *EU KLEMS* database. Second, following Acemoglu and Restrepo (2017), we control for whether a given industry is declining in terms of change in the nationwide employment levels between 1993 and 1998. Finally, we introduce control variables for changes in the capital intensity, and ICT capital in our regressions, respectively. We obtain information on the capital stock (i.e., net capital stock volume in millions) for each industry from the OECD's STAN database, and calculate the percentage change in the capital stock between 1999 and 2007. The change in the ICT capital is calculated analogous to the change in the capital intensity variable. The information on industry-level ICT capital is collected from the *EU KLEMS* database. In Panel D of Table II, we provide some descriptive statistics for the industry controls.

3 Empirical Specification

In our empirical analysis, we investigate the long-term relationship between changes in exposure to robots and changes in economic outcomes of individuals, accounting for a wide range of household characteristics, relevant industry trends, and local economic conditions through regional fixed effects. Our base model takes the following form:

$$\Delta Y_{ijk}^{99 \to 07} = \alpha \cdot \Delta Robot_Density_{j}^{99 \to 07} + \beta \cdot HH_Controls_{i}^{99} + \gamma \cdot \Delta IND_Controls_{j}^{99 \to 07} + \delta_{k} + \epsilon_{ijk}$$

$$\tag{4}$$

where $\Delta Y_{ijk}^{99 \rightarrow 07}$ represents the long-differences (that we refer to changes between 1999 and 2007) in the economic and financial outcomes of interest for household *i* who works in industry *j* and lives in municipality *k* in 1999.

Our dataset provides detailed and highly accurate information on numerous household demographic and financial characteristics, which are represented by vector $HH_Controls_i^{99}$. In our regressions, we control for household age and age squared to capture the hump-shaped profile of wealth and financial decisions, educational attainment measured by three different indicator variables (i.e., less than high school, high school graduate, and college and more), gender, marital status of the household head, separate variables for number of adults and number of children in the household, a dummy variable for whether the household is an immigrant, (inverse hyperbolic sine of) household disposable income, and quartile dummies for household net wealth. Note that all household controls are defined for the initial time period, which is 1999.

In addition to household characteristics, we account for several relevant industry controls (i.e., $IND_Controls_j^{99\to07}$) so as to isolate the effect of automation from other industry level trends and changes. The vector $IND_Controls_j^{99\to07}$ includes changes in exposure to Chinese imports between 1999 and 2007; percentage changes in capital intensity and ICT capital; and change in the nationwide industry level employment between 1993 and 1998, all of which are described in the previous section. Also, in additional robustness exercises, we account for contemporaneous median changes in import exposure to Western European countries to address the potential effects of international competition on the economic outcomes of sampled households. Lastly, we control for regional fixed effects, defined at the municipality level and denoted by δ_k , to account for potential differences in regional economic conditions and environment. In Sweden, there are a total 290 municipalities, which are responsible for various tasks such as social services or physical planning. Hence, the municipality fixed effects account for possible latent regional characteristics and capture the direct effects of the location of households.¹⁷ In our main specifications, we also control for initial robot density (measured in our base year that is 1995) of an industry, which allows us to focus on the variation coming from differences in changes in robot density across industries within a municipality.

Even though we include a rich set of industry controls in our regressions, there may be still some unobserved changes and trends that can be correlated with growth in robot density and wages and employment prospects in that industry. This would pose a threat to our identification. For example, a rapid increase in unionization in the Swedish industries could both lead to an increased adoption of robots and higher wages and improved job security in those industries, which would yield a positive correlation without implying a causal link between the two.¹⁸

To pin down the causal effects of increased exposure to robots on economic outcomes of households, we use an instrumental variable (IV) approach that is estimated in a 2SLS fashion. Following a similar identification strategy as in Autor et al. (2013), Bloom et al. (2016), and Acemoglu and Restrepo (2017), we instrument for changes in robot density between 1999 and 2007 in the Swedish industries using the contemporaneous median changes in robot density across eleven other developed Western European countries.¹⁹ Building on the same ideas as in Acemoglu and Restrepo (2017), we use adoption of robots in the (non-Swedish) European countries to capture the advances in global technological frontier, which is assumed to be positively correlated with the robot density growth in Sweden but uncorrelated with the error term in the equations of interest.²⁰ Indeed, the first-stage regressions show a positive and statistically highly significant effect (*p*-value<0.01) of the excluded in-

¹⁷Also, note that municipalities represent the most disaggregated geographic level in Sweden for which macroeconomic information is collected and calculated.

¹⁸Note that if any unobserved industry shocks are positively correlated with the degree of automation and employment and other economic outcomes in that industry, OLS regressions would underestimate the true effect of automation on household economic behavior. In other words, the OLS coefficients on the exposure to robots variable would be downward biased.

¹⁹The eleven other developed countries are Austria, Belgium, Denmark, Finland, France, Germany, Italy, Netherlands, Spain, Portugal, and the United Kingdom.

²⁰To the latter, we are unable to perform any over-identification tests such as the Hansen-J statistic to test the statistical exogeneity of the excluded instrument since our model is exactly identified.

strument on the endogenous robot exposure variable. In addition, we observe that the Fstatistics for the first-stage regressions are far greater than 10, which indicates that the
excluded instrument is strongly correlated with the endogenous robot exposure variable
and thus do not suffer from a weak instrument problem.²¹

Overall, our IV strategy will identify the exogenous variation in robot adoption in the Swedish industries that is induced by advances in the technological frontier of robotics, and further allows us to isolate the effect of an (exogenous) increase in robot density on the economic outcomes of households.

Finally, recall that Equation 4 is defined and estimated in first differences (i.e., changes between 1999 and 2007). Also, it is important to note that we correct standard errors for potential spatial correlation across households within a municipality by clustering at the municipality level. We obtain similar results when we account for possible serial correlation and heteroskedasticity within municipality-industry level (i.e., there is a total of 2,942 municipality-industry cells), which we present in the Online Appendix.

4 The Effect of Robots on Earnings and Unemployment Risk

We begin our empirical analysis with investigating the relationship between labor market outcomes of households and changes in their exposure to robots at work by considering two

²¹As in almost every IV estimation, we acknowledge that our identification strategy may be subject to some potential limitations. For example, the IV estimates may be contaminated if there is a negative correlation between the excluded instrument, i.e., changes in robot exposure in the (non-Swedish) European countries, and changes in the economic outcomes of Swedish households. Such a negative correlation may arise if any negative shock to the Swedish industries (that affect the labor market and financial outcomes of the Swedish workers) is positively correlated with the adoption of robots in other advanced countries. It is however important to note that correlations between our excluded instrument and other industry level trends and changes are relatively weak, ranging from 0.06 to 0.2. Still, to overcome, or at least to minimize, the severity of this problem, we account for various industry level trends in Sweden such as the contemporaneous changes in exposure to Chinese imports, percentage changes in the capital intensity and IT capital, import exposure to imports from European countries that we consider to construct the excluded instrument, and early trends in employment growth and adoption of robots of a given industry in all our regressions.

important dimensions of labor market outcomes. Specifically, we analyze whether working in an industry that experienced a higher robot adoption over the observation period affects the income growth and job-loss risk of households in that industry. Our analysis of the effect of robots on labor market parallels two recent pioneering papers that analyze the effects of penetration of robots on productivity, aggegate employment and wages (Graetz and Michaels, 2017; Acemoglu and Restrepo, 2017).²²

Table III presents the regression results on labor market outcomes.²³ In columns (1) and (2) of Table III, we consider the effects of increased automation on changes in house-hold income, which is defined as the log differences in earnings (net of any transfers or capitals gains) between 1999 and 2007. We find that both the OLS and the IV coefficients on robot exposure variable are negative and statistically significant, albeit, the IV coefficient is larger in magnitude, -0.009 versus -0.016. More specifically, based on the IV coefficient, a one-standard deviation exogenous rise in the robot density between 1999 and 2007, which corresponds to an increase of 3.27 robots per thousand workers, reduces the income growth of individuals by 5.2% (*t*-statistics: 2.60). This effect is economically meaningful, given that the average (nominal) income growth between 1999 and 2007 accounts for 19%.²⁴

²²Using the same dataset on industrial robot stock from the IFR, Graetz and Michaels (2017) analyze the effects of automation on labor productivity in a sample of 17 countries, and find that gains in labor productivity are significantly more pronounced for those country-industry pairs that experienced a higher adoption of robots between 1993 and 2007. The authors also show that robots reduce the employment share of low-skill workers, albeit, they find no effects on aggregate employment shares. In an another key contribution, Acemoglu and Restrepo (2017) analyze the equilibrium effects of automation on aggregate wages and employment shares across US commuting zones. Specifically, the authors first formally model the interactions between robots and workers in the production of different tasks, and derive testable predictions on how robots can affect aggregate labor market outcomes. They find that each additional robot per thousand workers decreases the aggregate employment rate by 37 and aggregate wages by 73 basis points, respectively.

²³A key distinction of our labor market analysis from the existing papers is that we use an extensive individual-level panel data and focus on the direct effects of automation on labor market outcomes of house-holds who are exposed to robots at work rather than the aggregate impact of robots. Furthermore, our dataset enables us to track the economic outcomes of the same individuals over a nine year period. Hence, both the panel dimension and rich nature of our dataset allow us to control for a wide range of individual characteristics and to isolate the effects of automation from other industry-level trends and region specific macro shocks.

²⁴In an additional test, we re-define the dependent variable, including the transfers received by the households such as unemployment benefits or social welfare payments for the alleviation of poverty in the income definition, and consider the effects of automation on the log differences in household disposable income (that is net of taxes and excludes capital income) between 1999 and 2007. The results, tabulated in Table O.A.2 in the Online Appendix, show that the effects of robots remain statistically significant while, not surprisingly, they decline in magnitude. This finding indicates that the negative contribution of automation to household

Next, we turn to the impact of automation on the unemployment risk of households. Our dependent variable is now an indicator variable that takes the value of 1 if a given household was employed in 1999, and is unemployed in 2007. In other words, we estimate the transition probability from employment to unemployment during the observation period. As noted by Fagereng et al. (2017), unemployment risk represents one of the most important sources of background risk - a risk that is non-tradable and not fully insurable due to market illiquidity or incompleteness (Kimball, 1993; Aiyagari, 1994; Heaton and Lucas, 2000).

Columns (3) and (4) of Table III report the estimation results. The IV estimates indicate that, ceteris paribus, a one-standard deviation increase in the robot density increases the probability to become unemployed by 1.5 percentage points. The effect is statistically highly significant (*t*-statistics: 5.52), and meaningful in economic terms. To put this into context, we compare the effect of automation with that of educational attainment. Our estimates imply that a one-standard deviation increase in robot density would fully offset the effect of having a high school degree (as compared to being a high-school dropout) on becoming unemployed, indicating that the impact of automation on job-loss risk is significant in magnitude. The positive contribution of robots to unemployment probability is consistent with Kogan et al. (2018) who document that advances in production methods are associated with substantial increases in the labor income risk of individual workers.

It is also worth briefly discussing some of the other controls in the regressions and their importance. First, and not surprisingly, we find that increased exposure to Chinese imports reduces the income growth and increases the unemployment probability of households. The negative and significant effect of exposure to imports from China on wages and employment is consistent with the findings of the prior literature (Autor et al., 2013; Bloom et al., 2016). More interestingly, we observe that changes in ICT capital in a given industry is positively correlated with the income growth. This result is in line with the findings of Acemoglu and Restrepo (2017) who note that the effect of robots on wages may be distinct from that income seems not to be fully offset by the transfers from the government. of ICT capital and other related information technologies (Acemoglu and Restrepo, 2017). When we turn to household controls, we observe that educational attainment and being a male positively contribute to income growth and employment, while immigrants tend to experience a lower income growth and face a higher job-loss risk than the natives during the observation period.

In summary, we establish negative and significant effects of robots on labor market outcomes of households. Next, we go beyond the labor market and investigate the effects of robots on household wealth accumulation.

5 Beyond the Labor Market: The Effect of Robots on Household Wealth

This section presents the results of household wealth regressions, and further discusses and explores mechanisms.

5.1 Robots and Distribution of Wealth

We next analyze the effects of automation on both static and dynamic aspects of the wealth distribution. Our main variable of interest is household net wealth, which we compute, as standard in the literature, by subtracting the household debt (i.e., mortgage loans and consumer debt) from total gross wealth that is the sum of all financial (i.e., direct and indirect stocks, bonds, cash) and real assets (i.e., value of primary residence, and other real estate holdings).

We measure the percentile rank of a household within the birth cohort-year distribution of net wealth, and use it as our preferred specification (Black et al., 2015; Chetty et al., 2014).²⁵ This wealth measure, by definition, accounts for life-cycle differences across house-

 $^{^{25}}$ We define 12 birth cohorts, with each birth cohort consisting of five-year intervals between the years 1923 and 1983.

holds (Black et al., 2015). Also, it can be defined for zero or negative values of net wealth, which is, for example, not feasible with a log transformation. Finally, when measuring the percentile rank of sampled households within their birth cohort distribution, we no longer restrict our attention to the households in the final sample but rather consider all households in the LINDA dataset with non-missing wealth information.

Before presenting the formal econometric estimates, we first provide some suggestive evidence on the relationship between increased use of robots and household wealth. In Figure I, we depict the dynamics of wealth dispersion, measured by the interquartile range of net wealth, between 1999 and 2007. Specifically, we sort the sampled households into three groups based on the automation growth of their industry of employment during the sample period, and compute the interquartile ranges of the net wealth for each group in each year.²⁶ As illustrated in the graph, all three groups exhibit increased wealth dispersion throughout the observation period. Interestingly, the rise in wealth dispersion monotonically increases in the changes in robot density, indicating potential effects of automation on household wealth accumulation. While Figure I is suggestive, we now formally analyze the relationship between automation and household wealth dynamics.

First, we use the net wealth percentile rank of a household in the end of the observation period as our dependent variable, and ask whether increased use of robots in an industry affects the relative position of households in the wealth distribution. Columns (1) and (2) of Table IV present the regression results. Both OLS and IV estimates imply a negative and statistically highly significant effect of exposure to robots on the rank of households in the wealth distribution (t-statistics: -8.15). To give an idea of the magnitude of automation effects, a one-standard-deviation increase in the robot density in an industry reduces the rank of individuals in the wealth distribution by 2.5 percentiles. To put this into context, we again contrast it with the effect of educational attainment level on the wealth rank of households. We observe that the effect of robots is equivalent to one-fifth of that of having

²⁶For comparison, we normalize these values by the initial IQR value.

a college degree (where high school drop-outs being the reference group). Hence, the magnitude of the automation effects is quite considerable, given the importance of educational attainment in the wealth accumulation (Huggett et al., 2011; Epper et al., 2018).

The percentile rank variable focuses on the net asset holdings of households and reflects the position of households in the wealth distribution relative to their peers. However, it is silent on the changes within the distribution of wealth over time (Quadrini, 2000). To address this issue, we next focus on the wealth mobility of households over the sample period. In particular, we define an indicator variable that is equal to 1 if a household falls into a lower wealth percentile rank within the birth cohort distribution in 2007 relative to her initial position in 1999, and zero otherwise. Note that this measure enables us to assess the intracohort mobility of households over time, and thus, provides insights about the potential effects of robots on the household wealth dynamics (net of any life-cycle effects).

Columns (3) and (4) of Table IV present our OLS and IV estimates. We find that households who are more exposed to industrial robots at work also experience greater downward mobility in the 1999-2007 period. The IV estimate in column (4) of 0.004 indicates that a one-standard-deviation exogenous rise in the robot use in an industry leads to a 1.2 percentage points increase in the probability of a household falling in the wealth distribution over the sample period (*t*-statistics: 2.41). To complement the analysis, we also consider the upward mobility of individuals, that is, the probability of moving to a higher wealth class during the observation period. In tests presented in Table O.A.3 in the Online Appendix, we document significant and negative effects of robots on upward mobility of individuals in the wealth distribution. All in all, the results from the mobility regressions are consistent with the patterns observed in Figure 1, indicating that rapid automation can contribute to a more dispersed wealth distribution.

Controlling for other household variables, we find that educational attainment and household income both negatively contribute to downward mobility, while they have a positive effect on the household rank in the wealth distribution. Interestingly, we observe that immigrants tend to be in lower ranks of the wealth distribution, however, they do not experience any downward mobility relative to natives during the observation period.

To address any concerns that our findings may be driven by the differences in the housing investments of households, in a robustness exercise, we verify our results including homeownership of households as an additional covariate in the wealth regressions. The results, tabulated in Table O.A.4, show that the negative impact of robots on household wealth remains almost identical even after conditioning on homeownership. Also, one can argue that observed variation in household wealth could be induced by differences in risk preferences of households. Table O.A.5 hence repeats the exercise controlling for initial risk exposure that is measured by the share of financial wealth in risky assets in 1999 (Fagereng et al., 2018). We again find very similar results. Finally, we analyze whether changes in household debt contribute to observed differences in household net wealth. For example, in a recent paper, Barrot et al. (2018) show that households who live in regions where manufacturing industries are more exposed to import competition significantly lever up to smooth their consumption. To address this possibility directly, we next regress the log changes in household debt between 1999 and 2007 on the industry-level changes in robot density over the same time period and other household and industry controls. As presented in Table O.A.6, we find no significant effect of increased robotization on household debt, suggesting that automation appears to affect household wealth through its effects on the asset side of household balance sheets. Collectively, wealth analysis yields strong evidence for the negative effects of pervasive automation on both static and dynamic aspects of wealth distribution.

5.2 Understanding the Mechanism

In what follows we discuss and explore mechanisms through which increased use of robots at work can affect the distribution of wealth.

A natural explanation for the negative wealth effects of automation could be the differences in earned incomes of households. Recall our findings that individuals who are working in industries with a higher rate of robot adoption on average experience lower income growth. Accordingly, wealth accumulation behavior of households could differ from each other, with those who are less exposed to robots accumulating higher levels of wealth over time.²⁷ To address this concern, we rerun our wealth regressions by including (the inverse hyperbolic sine of) household disposable income in 1999 and realized income growth between 1999 and 2007 as additional regressors in our analysis, both of which enter positively and significantly in the wealth regressions. As presented in Table O.A.7, we still observe that the exposure to robot variable remains significant with a negative sign. We obtain similar results even when we use the average disposable income over the 5 year period prior to 1999 in lieu of income levels in 1999 in these regressions (Guvenen et al., 2017).²⁸

A further potential consideration is that observed differences in wealth can arise from the heterogeneity in saving behavior across households from different industries, as argued by the standard models (De Nardi and Fella, 2017).²⁹ According to this explanation, individuals who are employed in more automated industries may have systematically lower active saving rates than those who work in industries with a lower rate of robotization. Hence, they end up in relatively lower ranks of the wealth distribution, and experience greater downward mobility.

We address this alternative explanation in a number of ways. First, we calculate the initial total saving rate of each sampled household and include this variable as an additional covariate in the wealth analysis.³⁰ The results, tabulated in Table O.A.8 and O.A.9, show

²⁷To illustrate this channel, consider two identical households owning the same initial level of wealth, earning the same (initial) income, and having identical savings rate. Under the assumption of homogeneous rates of returns to savings, as households from different industries experience different income growth due to robots, the one household that is less exposed to automation would accumulate higher levels of wealth over time.

²⁸Note that Guvenen et al. (2017) use the 5-year average income as a proxy for permanent income in their analysis. By the same logic, one can argue that the negative wealth effects of robots that we document are robust to controlling for differences in permanent income across people.

²⁹The existing theories argue that there can be various reasons for the heterogeneity of saving behavior. For example, the variation in the saving rates can be related to the heterogeneity in time preferences (Krusell and Smith, 1998) or to the differences in the levels of household wealth (Carroll, 1998).

³⁰We calculate the total saving rate of a household in the following way: We first calculate the annual differences in the household net wealth, and scale it by the household net wealth from the prior year. Following Bach et al. (2017), we only consider those households with non-zero or non-negative net wealth in the analysis.

that exposure to robots variable still exhibits a negative significant effect even after controlling for heterogeneity in saving behavior. We also verify these results using the initial active saving rate in lieu of total saving rates as shown in Table O.A.10. In another robustness exercise, we include initial wealth of households (as of 1999) in our wealth regressions. This analysis is motivated by Bach et al. (2017) who document a negative robust correlation between active saving rates of households and their wealth levels in Sweden. As presented in Tables O.A.11 and O.A.12, both the OLS and IV estimates indicate a negative significant effect of increased automation on household wealth variables. This finding is robust to measuring the initial wealth either in levels or using quartile wealth dummies.

Another possible mechanism for our findings is the heterogeneity in the saving-induced growth component of wealth accumulation.³¹ In principle, differences in new savings across households have the potential to explain the variation in wealth (Saez and Zucman, 2016). In our context, however, this explanation would require households who are less exposed to robots at work to increase their new savings (as compared to those who are more exposed to robots) to generate the observed negative relationship between automation growth and wealth accumulation. This is the opposite of what theory predicts. For example, models of precautionary saving suggest that (prudent) households would accumulate more assets when they are confronted with greater income uncertainty (Kimball, 1990; Carroll, 1998; Lusardi, 1998), which implies an increase in the new savings of households with a higher (not lower) exposure to robots. Still, to try to control for this possibility, we reestimate our wealth regressions controlling for the average active saving rate of each sampled household between 2000 and 2007.³² We show in Table O.A.13 that the results are qualitatively

Finally, we winsorize the saving rate variable at 1% level to eliminate any concerns about the outliers. We also compute the total saving rates by scaling the total savings by current household income.

³¹Piketty and Zucman (2014) argue that wealth growth can be decomposed into a saving-induced growth component (i.e., volume effects) and a capital-gains-induced growth component (i.e., relative price effects).

³²To calculate the active saving rates, we make use of an auxiliary dataset provided by Statistics Sweden, which includes individual-level security information on the portfolio holdings of each (and every) individual in the LINDA dataset. Using standard financial databases (*inter alia* Bloomberg, Thomson Reuters Datastream, Factset, Compustat, Thomson Reuters Mutual Funds), we next collect end-of-year adjusted prices for (sampled) single stocks and mutual funds, and calculate their annual raw returns. We then compute the one year value-weighted buy-and-hold returns for each household's equity portfolio using the weighted-sum of the

very similar to the base results. Taken together, our results suggest that wealth differences across households appear not to be merely a consequence of income differences or heterogeneity in saving behavior of individuals across industries.

Lastly, increased automation can affect wealth accumulation through its potential effects on financial risk taking and investment choices of households. Even though early models of wealth inequality assumes homogeneous rates of return (Bewley, 1977), recent empirical literature documents considerable heterogeneity in returns to wealth (Bach et al., 2016; Fagereng et al., 2018), which emerges as an important channel to explain the variation in household wealth (Benhabib et al., 2011; Gabaix et al., 2016; Benhabib et al., 2017). We thus turn to examine this potential mechanism.

5.3 Robots, Financial Risk Taking, and Financial Wealth

How can increased automation affect returns to wealth? The rapid adoption of robots at work leads individuals to face higher background labor income risk, as shown by our labor market analysis in Section 4. The theory argues that increased background risk and its associated costs reduce the willingness of investors to take other types of risk, such as holding risky financial assets (Eeckhoudt et al., 1996).³³ As returns to wealth are directly affected by willingness of households to take financial risk (Ameriks et al., 2003), reducing or completely eliminating the exposure to stock market (in response to increased human capital risk) would lead to accumulating lower levels of wealth over time, which is also sup-

portfolio share of each stocks or mutual fund from the prior year and their annual returns. Here, we assume that all portfolio inflows and outflows occur at the end of each year. Using this information, we then decompose the changes in the financial wealth into two components: (i) active changes (i.e., due to new savings), and (ii) passive changes (i.e., due to returns on risky investments). Finally, we calculate the active saving rate of a given household a given year by dividing the active changes in the financial wealth by the contemporaneous disposable household income. To alleviate any concerns about the outliers, we winsorize this variable at 1%.

³³In a key contribution, Cocco et al. (2005) build and simulate a life-cycle model of consumption and portfolio choice with non-tradable labor income, and show that individuals who are exposed to more idiosyncratic labor income risk invest less in stocks. They also estimate the welfare losses incurred by ignoring labor income when investing in risky assets, and find them to be up to 2% of annual consumption of investors. The recent empirical literature also provides evidence that is consistent with the predictions of the theory (Betermier et al., 2012; Fagereng et al., 2017). For example, using a similar administrative dataset to ours, Fagereng et al. (2017) find that individuals respond to increased labor income risk by reducing their financial risk exposure, with the effect being more pronounced among less wealthy households.

ported by the data. For example, using household data from Sweden, Bach et al. (2016) find that individuals in the top 1% of the wealth distribution earn 400 basis higher annual returns on their financial wealth than the median household, which, as the authors argue, are primarily compensations for their exposure to higher levels of systematic risk.³⁴

Given this background, we now turn to analysis of the effects of exposure to robots on financial risk taking of households and financial wealth accumulation, respectively. Table V reports our OLS and IV estimates. In Columns (1) and (2), the dependent variable is the stockholding status of a given household that is an indicator variable, which takes the value of 1 if household i holds directly or indirectly stocks in 2007. In other words, we first focus on the static aspect of the household financial behavior, and estimate the probability of *being* a stockowner.

Even after controlling for endogeneity of the robot exposure variable and other wellknown predictors of stockholding, we find that increased exposure to robots in an industry reduces the probability of households in that industry to hold stocks, although the IV coefficient is smaller in magnitude and only marginally significant (*t*-statistics:-1.83). In terms of economic magnitude, ceteris paribus, a one-standard-deviation increase in the robot density lowers the likelihood to be a stockowner by approximately 1 percentage point. In fact, both economically and statistically weaker results using the stockholding status as a dependent variable is somewhat consistent with the previous findings of the literature. For example, in their analysis of the effect of labor income risk on asset allocation, Betermier et al. (2012) find weaker effects when they consider level of risky share rather than the changes in risky share. The authors attribute this difference to the cross-sectional unobserved "taste" differences.

To address this concern, in Column (3) and (4), we next consider changes in stockholding behavior by focussing on the exit decisions of households from the stock market, shifting our

 $^{^{34}}$ In another important contribution, Fagereng et al. (2018) document a return spread of 260 basis points between the 90th and 10th percentile of the financial wealth distribution in Norway. See also Campbell et al. (2018) who document that heterogeneity in returns to equity wealth contributes to increased equity wealth inequality in India.

analysis from the static aspect of household financial behavior to its dynamics. In particular, the dependent variable is now a dummy variable that equals to 1 if household i held both directly and indirectly stocks in 1999 but liquidiated all her stock holdings in 2007. We only include those households who remain in the stock market throughout the sample period in the reference group, which reduces the sample size by approximately 8,250 to 22,125 households.

Consistent with the theoretical expectations, our results indicate that an increase in the robot density in an industry increases the probability of stock market exit. The effect is not only statistically significant (t-statistics: 2.43) but also economically highly meaningful: the IV estimate imply that a one-standard-deviation exogenous rise in the robot use growth, that again corresponds to a 3.27 robot increase per thousand workers between 1999 and 2007, increases the likelihood of exiting from the stock market by approximately 1 percentage point. The magnitude of the coefficient estimate might appear at first glance relatively small in economic terms. Though, this estimate implies an 11.4% increase in the exit probability, as the stock market exit rate in our sample equals to 8.20%. It is important to note that our results are robust to controlling for unobserved household heterogeneity as reported in O.A.18. Overall, this finding conforms to our proposed mechanism that increased adoption of robots at work, and hence, the higher human capital risk makes households less willing to take financial risks. Consequently, investors fully rebalance their financial portfolio away from stocks, and forego substantial equity returns up to 4.3% on a year by not participating in the stock market (Calvet et al., 2007).³⁵

The sign and statistical significance of other investor controls in our estimation model is consistent with what has been found in household finance regressions for these vari-

³⁵As a potential policy recommendation, Freeman (2015) argues that being a capital owner (e.g., either by directly or indirectly - through private pension funds or mutual funds - investing in the stocks of companies that produce or employ robots that can substitute for human workers) could limit the adverse impact of automation on economic well-being of individuals if they were to earn a higher share of their income from capital ownership. However, our findings indicate that automation not only exerts direct downward pressure on wages of individual workers, but also prevents them to receive a share of potential productivity gains by discouraging individuals to invest in the stock market.

ables to date (Brunnermeier and Nagel, 2008). For example, exit probability monotonically decreases in household wealth. Similarly, higher levels of educational attainment and disposable household income also negatively contribute to the decision of households to exit from the stock market.

Motivated by our findings on financial risk taking, we finally investigate whether automation also contributes to differences in accumulation of financial wealth across households. To do so, we first regress percentage changes in financial wealth (defined as the log differences in financial wealth between 1999 and 2007) on the changes in robot density over the sample period, and other household and industry controls.³⁶ As presented in columns (1) and (2) of Table VI, we observe that increased exposure to robots at work indeed negatively contributes to the changes in household financial wealth between 1999 and 2007. In terms of magnitude, based on the IV coefficient, a one-standard deviation exogenous increase in the robot density reduces the rate of financial wealth accumulation by 15.3% over the observation period (t-statistics:-4.71). Finally, we consider the effects of increased automation on the financial wealth to income ratio of individuals in 2007.³⁷ Column (3) and (4) of Table VI report the estimation results. Both the OLS and IV coefficients on changes in the robot density variable is negative and statistically highly significant (t-statistics:-5.85), suggesting that households who are working in industries with a higher rate of robot adoption end up accumulating less financial wealth relative to their income over time.³⁸ Taken as a whole, these findings support the notion that individuals who eliminate their exposure to stock market in response to increases in their human capital risk experience a substantial

³⁶Financial wealth is defined as the sum of the value of direct and indirect stocks, bonds, bond and mixed mutual funds, and cash holdings in the savings and checking accounts. Note that percentage change in financial wealth can be either due to passive changes (i.e., returns on financial wealth) or active changes. Finally, it is important to mention that we winsorize this variable at the 1 percent level.

³⁷We divide the financial wealth by the household earnings net any any transfers and capital gain. Again, we winsorize the financial wealth-to-income ratio at the 1 percent level to alleviate any concerns that our results might be driven by outliers.

³⁸In fact, models of precautionary saving imply that prudent households would accumulate more assets when they are confronted with greater income uncertainty (Kimball, 1990; Carroll, 1998; Lusardi, 1998). Our findings suggest that this effect (i.e., the precautionary saving motive) appears to be offset (or even reversed) by the relative price effects due to differences in the composition of financial portfolio and thereby generate an overall negative effect of automation on financial wealth.

drop in their financial wealth growth and accumulate less financial wealth.

6 Distributional Effects of Robots

Up until now in our analysis, we have worked under the assumption that all employees within an industry are to a similar degree affected by increased use of robots in that industry. However, it is conceivable to argue that automation can have differential effects on the economic and financial outcomes of households, depending on the required skill-level or type of their occupations. Consistent with this conjecture, Acemoglu and Restrepo (2018b) show that middle-aged workers who perform blue-collar tasks are more likely to be replaced by industrial robots relative older workers who are specialized in non-production services.

To study the distributional effects of automation, we next focus on the economic behavior and wealth accumulation of households by skill-level. Following the standard in the literature (see e.g., Card and Lemieux, 2001; Acemoglu et al., 2004), we use the level of educational attainment of households, and define a low-skill group that corresponds to either being a high-school graduate or less, and a high-skill group that includes households with a college education and more. We then rerun our benchmark model given in Equation 4 for these groups separately. We argue that education level of households should serve as a good proxy for their skill-level, with the less-educated being more likely to perform blue-collar tasks, and hence, more prone to the effects of robots.

Table VII presents the key results for our cross-sectional analysis.³⁹ In Panel A, we first focus on the labor market outcomes of households. As shown in columns (1) and (3), we find that less-educated households indeed experience a greater decline in income growth and a higher employment risk due to the increased number of industrial robots in their industry of employment. Interestingly, negative wage and employment effects are also present for the better-educated households, albeit, the estimates are only marginally significant and

³⁹Note that we present the results of the IV regressions, which represents our preferred estimation model. For brevity, we only report the coefficient estimates on exposure to robot variable, but we of course control for all relevant household characteristics, industry controls, and regional fixed effects in our regression.

decline in magnitude. The latter result is somewhat surprising but consistent with the findings of Acemoglu and Restrepo (2017) who argue that industrial robots appear not to complement any particular skill-group of workers, unlike other types of recent technologies such as computerization (Autor and Dorn, 2013).

When we analyze the wealth effects by skill-level, an interesting pattern emerges. As presented in Panel B of Table VII, we only observe negative wealth effects for low-educated households, who experience a greater probability of moving to lower wealth classes and ending up in a lower rank in the wealth distribution. On the other hand, we find no significant effect on both static and dynamic aspects of wealth accumulation for better-educated households. A similar perspective also applies to household financial risk taking. As presented in Panel C, for the low-educated households, the IV coefficient on changes in robot density is negative and significant at the one-percent level, while it is not statistically different than zero for households with at least a college degree. Accordingly, households in the less-educated group experience a substantial drop in the growth of their financial wealth, whereas we find no significant effect of robots on changes in financial wealth among the better-educated households.⁴⁰

In summary, there appears to be asymmetric wealth effects of robots across different segments of the population, which can have important implications. For example, expert opinion suggests that global robot stock could reach to three to four times larger levels (relative to its current level) over the next decade (see for example Acemoglu and Restrepo, 2017 and the references therein). According to our results, such a rapid growth in robotics could contribute to widening, the already large and persistent, wealth gap across households with different educational attainment levels, which can in turn create distributional challenges in the future.

⁴⁰Also, it is also important to note that the increase in industrial robot use has a more pronounced negative effect on the financial wealth accumulation of the less-educated group.

7 Additional Robustness and Sensitivity Analysis

In this section, we perform several additional tests to ensure the robustness of our findings. We present these results in the Online Appendix. First, we verify our results excluding from the sample those individuals who are working in the automative industry, which has historically the highest robot density per thousand workers in Sweden. The results, tabulated in Table O.A.14, are consistent with those of our baseline regressions, suggesting that our results are not merely driven by the automative industry.

To further investigate the robustness of our results, we next eliminate individuals working in the rubber and plastic industry that experienced the largest growth in robot use across industries in Sweden during the observation period. As presented in Table O.A.15, we find qualitatively and quantitatively similar results even after the exclusion of this industry from the sample, suggesting that our results are not affected by any outliers.

Next, we address the potential sorting of households into different industries. In particular, individuals may anticipate the increased adoption of industrial robots at their workplace in the beginning of the observation period. Hence, they may sort themselves into sectors that have a lower potential for increased use of robots. The non-randomness in the sorting of individuals to different industries could bias the coefficient estimates on exposure to robots variable. To address this concern, we next repeat our benchmark analysis focusing only on households who have been employed in the same industry since 1995 or earlier (i.e., when concerns over robotization have not yet gained much prominence). This restriction reduces the sample size by approximately 11,200 from 30,375 to 19,178 households. As shown in Table O.A.16, we observe that our findings are robust to basing our exposure variable on the households' industry of employment from the prior decade, indicating that endogenous selection of individuals to different industries appear not to drive our results. In untabulated tests, we also verify these findings using industry of employment information from 1993.

To ensure that our findings are robust to the effects of international competition from

the European countries, in our IV regressions, we also control for contemporaneous median changes in import exposure to eleven developed Western European countries (that we use to construct the excluded instrument). The results, tabulated in Table O.A.17, show that exposure to robots variable still exhibits a negative impact on economic outcomes of households and has a distinct effect from that of the international competition.

In an additional robustness exercise, we remove unobserved time-invariant household heterogeneity by first-differencing the wealth and stock market models. The regression results of the first-differenced models are reported in Table O.A.18. When controlling for household fixed effects, we observe that the exposure to robot variable remains its significance in all regressions. It is also important to note that the magnitude of the coefficient estimates increase drastically when we control for fixed effects.

Lastly, we consider an alternative definition of the technological frontier and rerun the IV regressions. In particular, we use the median changes in robot density per thousand workers in the Northern European countries, which are Austria, Belgium, Denmark, Finland, Germany, and the Netherlands. We note that the results, presented in Table O.A.19, are qualitatively similar to our baseline results.

8 Conclusions

This paper uses an extensive administrative panel and auxiliary data on stock and new installations of industrial robots to estimate the effect of increased automation on household wealth dynamics. We find evidence of statistically and economically significant effects of spread of industrial robots on static and dynamic aspects of wealth distribution. In particular, exposure to robots reduces the percentile wealth rank of an individual within her birth cohort-year distribution, and significantly increases the probability of downward mobility between 1999 and 2007. Our findings are robust to correcting for the endogeneity of exposure to robots, and controlling for a rich set of household characteristics, macroeconomic and institutional regional factors, as well as a range of industry trends.

We consider a number of alternative explanations for our results in the process of exploring the mechanism through which industrial robots can affect household wealth accumulation. We show that differences in earned incomes or differential saving rates alone do not seem to explain the differences in levels and dynamics of household wealth. In addition, we argue and provide evidence that the negative impact of automation (through increasing uninsurable human capital risk) on financial risk taking and investment choices of individuals appears to be an additional channel, which contributes to differences in levels and dynamics of household wealth. We also find that the wealth effects of automation are only operative for the subsample of households with low levels of education. The asymmetric wealth effects of robots across low- and high-skill workers caution against distributional challenges of automation.

All in all, our findings suggest the presence of significant effects of automation that extend beyond labor market to the distribution of wealth, and contribute to the current discussion on the economic consequences of automation.

References

- Acemoglu, D., D. H. Autor, and D. Lyle (2004). Women, war, and wages: The effect of female labor supply on the wage structure at midcentury. *Journal of Political Economy 112*(3), 497–551.
- Acemoglu, D. and P. Restrepo (2017). Robots and jobs: Evidence from us labor markets. Technical report, National Bureau of Economic Research.
- Acemoglu, D. and P. Restrepo (2018a). Artificial intelligence, automation and work. Technical report, National Bureau of Economic Research.
- Acemoglu, D. and P. Restrepo (2018b). Demographics and automation. Technical report, National Bureau of Economic Research.
- Aghion, P., B. F. Jones, and C. I. Jones (2017). Artificial intelligence and economic growth. Technical report, National Bureau of Economic Research.
- Aiyagari, S. R. (1994). Uninsured idiosyncratic risk and aggregate saving. The Quarterly Journal of Economics 109(3), 659–684.
- Ameriks, J., A. Caplin, and J. Leahy (2003). Wealth accumulation and the propensity to plan. *The Quarterly Journal of Economics* 118(3), 1007–1047.
- Autor, D. and A. Salomons (2018). Is automation labor share-displacing? productivity growth, employment, and the labor share. *Brookings Papers on Economic Activity 2018*(1), 1–87.
- Autor, D. H. (2015). Why are there still so many jobs? the history and future of workplace automation. *Journal of Economic Perspectives 29*(3), 3–30.
- Autor, D. H. and D. Dorn (2013). The growth of low-skill service jobs and the polarization of the us labor market. *American Economic Review 103*(5), 1553–97.

- Autor, H., D. Dorn, and G. H. Hanson (2013). The china syndrome: Local labor market effects of import competition in the united states. *American Economic Review 103*(6), 2121-68.
- Bach, L., L. Calvet, and P. Sodini (2017). From saving comes having? disentangling the impact of savings on inequality.
- Bach, L., L. E. Calvet, and P. Sodini (2016). Rich pickings? risk, return, and skill in the portfolios of the wealthy.
- Barrot, J.-N., E. Loualiche, M. C. Plosser, and J. Sauvagnat (2018). Import competition and household debt.
- Benhabib, J., A. Bisin, and M. Luo (2017). Wealth distribution and social mobility in the us: A quantitative approach. Technical report, National Bureau of Economic Research.
- Benhabib, J., A. Bisin, and S. Zhu (2011). The distribution of wealth and fiscal policy in economies with finitely lived agents. *Econometrica* 79(1), 123–157.
- Benzell, S. G., L. J. Kotlikoff, G. LaGarda, and J. D. Sachs (2015). Robots are us: Some economics of human replacement. Technical report, National Bureau of Economic Research.
- Berg, A., E. Buffie, and L.-F. Zanna (2018). Should we fear the robot revolution?(the correct answer is yes). *Journal of Monetary Economics*.
- Betermier, S., T. Jansson, C. Parlour, and J. Walden (2012). Hedging labor income risk. Journal of Financial Economics 105(3), 622–639.
- Bewley, T. (1977). The permanent income hypothesis: A theoretical formulation. Journal of Economic Theory 16(2), 252–292.
- Black, S. E., P. J. Devereux, P. Lundborg, and K. Majlesi (2015). Poor little rich kids? the determinants of the intergenerational transmission of wealth. Technical report, National Bureau of Economic Research.

- Bloom, N., M. Draca, and J. Van Reenen (2016). Trade induced technical change? the impact of chinese imports on innovation, it and productivity. *The Review of Economic Studies* 83(1), 87–117.
- Brunnermeier, M. K. and S. Nagel (2008). Do wealth fluctuations generate time-varying risk aversion? micro-evidence on individuals. *American Economic Review* 98(3), 713–36.
- Brynjolfsson, E. and A. McAfee (2012). *Race against the machine: How the digital revolution is accelerating innovation, driving productivity, and irreversibly transforming employment and the economy.* Brynjolfsson and McAfee.
- Calvet, L. E., J. Y. Campbell, and P. Sodini (2007). Down or out: Assessing the welfare costs of household investment mistakes. *Journal of Political Economy* 115(5), 707–747.
- Campbell, J. Y., T. Ramadorai, and B. Ranish (2018). Do the rich get richer in the stock market? evidence from india. Technical report, National Bureau of Economic Research.
- Card, D. and T. Lemieux (2001). Can falling supply explain the rising return to college for younger men? a cohort-based analysis. *The Quarterly Journal of Economics 116*(2), 705–746.
- Carroll, C. D. (1998). Why do the rich save so much? Technical report, National Bureau of Economic Research.
- Chetty, R., N. Hendren, P. Kline, and E. Saez (2014). Where is the land of opportunity? the geography of intergenerational mobility in the united states. *The Quarterly Journal of Economics 129*(4), 1553–1623.
- Cocco, J. F., F. J. Gomes, and P. J. Maenhout (2005). Consumption and portfolio choice over the life cycle. *The Review of Financial Studies 18*(2), 491–533.
- De Nardi, M. and G. Fella (2017). Saving and wealth inequality. *Review of Economic Dy*namics 26, 280-300.

- Eeckhoudt, L., C. Gollier, and H. Schlesinger (1996). Changes in background risk and risk taking behavior. *Econometrica: Journal of the Econometric Society*, 683–689.
- Epper, T., E. Fehr, H. Fehr-Duda, C. T. Kreiner, D. D. Lassen, S. Leth-Petersen, and G. N. Rasmussen (2018). Time discounting and wealth inequality.
- Fagereng, A., L. Guiso, D. Malacrino, L. Pistaferri, et al. (2018). Heterogeneity and persistence in returns to wealth. Technical report, CESifo Group Munich.
- Fagereng, A., L. Guiso, and L. Pistaferri (2017). Portfolio choices, firm shocks, and uninsurable wage risk. *The Review of Economic Studies* 85(1), 437–474.
- Freeman, R. B. (2015). Who owns the robots rules the world. IZA World of Labor.
- Gabaix, X., J.-M. Lasry, P.-L. Lions, and B. Moll (2016). The dynamics of inequality. *Econo*metrica 84(6), 2071–2111.
- Graetz, G. and G. Michaels (2017). Robots at work. *Review of Economics and Statistics*, Forthcoming.
- Guvenen, F., S. Schulhofer-Wohl, J. Song, and M. Yogo (2017). Worker betas: Five facts about systematic earnings risk. *American Economic Review* 107(5), 398–403.
- Heaton, J. and D. Lucas (2000). Portfolio choice in the presence of background risk. *The Economic Journal 110*(460), 1–26.
- Huggett, M., G. Ventura, and A. Yaron (2011). Sources of lifetime inequality. American Economic Review 101(7), 2923–54.
- Hurst, E. and A. Lusardi (2004). Liquidity constraints, household wealth, and entrepreneurship. *Journal of Political Economy 112*(2), 319–347.
- IFR (2017). Wold robotics: Industrial robots. Technical report, International Federation of Robotics.

- Jäger, K. (2016). Eu klems growth and productivity accounts 2017 release, statistical module1. December, http://www. euklems. net/TCB/2016/Metholology_EU% 20KLEMS_2016. pdf.
- Kimball, M. S. (1990). Precautionary saving in the small and in the large. Econometrica: Journal of the Econometric Society, 53–73.
- Kimball, M. S. (1993). Standard risk aversion. Econometrica: Journal of the Econometric Society, 589–611.
- Kogan, L., D. Papanikolaou, L. Schmidt, and J. Song (2018). Technological innovation and the distribution of labor income growth. Technical report.
- Krusell, P. and A. A. Smith, Jr (1998). Income and wealth heterogeneity in the macroeconomy. Journal of political Economy 106(5), 867–896.
- Lusardi, A. (1998). On the importance of the precautionary saving motive. *The American Economic Review* 88(2), 449–453.
- Piketty, T. and G. Zucman (2014). Capital is back: Wealth-income ratios in rich countries 1700–2010. The Quarterly Journal of Economics 129(3), 1255–1310.
- Quadrini, V. (2000). Entrepreneurship, saving, and social mobility. Review of Economic Dynamics 3(1), 1–40.
- Sachs, J. D., S. G. Benzell, and G. LaGarda (2015). Robots: Curse or blessing? a basic framework. Technical report, National Bureau of Economic Research.
- Sachs, J. D. and L. J. Kotlikoff (2012). Smart machines and long-term misery. Technical report, National Bureau of Economic Research.
- Saez, E. and G. Zucman (2016). Wealth inequality in the united states since 1913: Evidence from capitalized income tax data. *The Quarterly Journal of Economics* 131(2), 519–578.

Figure I: Wealth Dispersion over Time

This figure depicts the dynamics of wealth dispersion, measured by the interquartile range of net wealth, between 1999 and 2007. Specifically, we sort the sampled households into three groups based on the automation growth of their industry of employment during the sample period, and compute the interquartile ranges of the net wealth for each group in each year. For comparison, we normalize these values by the initial IQR value.



Table I: Use of Industrial Robots in the Swedish Industries

This table presents descriptive statistics on the use of industrial robots in Swedish industries. In column (1), we report the number of sampled households who are working in the industries that we consider in our analysis. Column (2) presents the number of workers in thousands in each industry in 1995. Columns (3) and (4) present the number of industrial robots for 1999, and 2007, respectively. In columns (5) and (6), we report the robot density per thousand workers for 1999, and 2007, respectively. Finally, column (7) presents the changes in robot density between 1999 and 2007 for each industry separately. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS database.

Name of Industry	No of obs	No of Workers 1995	No of Robots 1999	No of Robots 2007	Robot Density 1999	Robot Density 2007	Change in Robot Density 1999-2007
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Agriculture, forestry, fishing	1,215	46	0	1	0	0.022	0.0217
Food and beverages; tobacco	1,699	69	96	416	1.391	6.029	4.637
Textiles	320	15	7	0	0.467	0	-0.467
Wood and furniture, Paper	3,863	112	82	189	0.732	1.687	0.955
Pharmaceuticals, cosmetics; other chemical products	$1,\!275$	42	0	80	0	1.904	1.905
Rubber and plastic products; chemical products	1,180	46	260	790	5.652	17.173	11.521
Basic metals; Metal products	$3,\!483$	110	1249	1943	11.354	17.663	6.309
Industrial machinery	3,142	88	551	576	6.261	6.545	0.284
Electrical/electronics	2,723	91	356	569	3.912	6.252	2.340
Automotive; Other vehicles	3,263	87	2424	3089	27.862	35.505	7.643
Education/research/development	1,666	398	129	93	0.324	0.234	-0.0904
Construction	$5,\!372$	187	39	49	0.209	0.2620	0.0534
Electricity, gas, water supply	878	41	1	1	0.024	0.024	0
Mining and quarrying	296	10	0	0	0	0	0

Table II: Summary Statistics For the Final Sample

This table presents the number of observations, mean, and standard deviation of variables used in the empirical analysis. In Panel (A), we present the descriptive statistics for the household control variables that are defined in 1999. In Panel B, we report the summary statistics for the main variables of interest in our analysis, that are the changes in robot density in Swedish industries and the median change in the 11 developed Western European countries that we use as an instrument. Panel C presents the descriptive statistics for the dependent variables that we consider in our empirical analysis, and Panel D report the industry controls. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS, and OECD's STAN databases.

	No of Obs	Mean	Std. Dev.				
	(1)	(2)	(3)				
Panel A. Household Demographics							
Age	30,375	38.8532	7.526128				
Male	30,375	.8698601	.3364628				
Married	30,375	.550749	.4974261				
College and more	30,375	.2424362	.4285638				
High school	30,375	.5581235	.4966183				
Nbr of adults	30,375	1.926782	.6272694				
Nbr of children	30,375	1.416823	1.149026				
Net wealth	30,375	524553.4	1730830				
(IHS) Disposable income	30,375	13.2065	.4074581				
(IHS) Labor income	30,375	12.71247	.5225056				
Immigrant	30,375	.0995226	.2993674				
Pa	nel B. Varia	bles of Interes	st				
$\Delta Robot_Density^{99 \rightarrow 07}$	30,375	2.692674	3.274397				
$\Delta Robot_Density^{99 \rightarrow 07}_{EU}$	30,375	.4225471	.5255308				
Pa	nel C. Depen	dent variable	28				
Stockholding status (2007)	30,375	.7808066	.4137068				
Exit from the stock market	22,125	.0819887	.2743537				
Transition to unemployment	30,375	.0423374	.2013612				
Change in log earnings	30,375	.1955055	1.547096				
Downward wealth rank mobility	30,375	.5279342	.4992273				
Net wealth rank (2007)	30,375	52.9843	27.16513				
Wealth-to-income ratio (2007)	29,955	.8727597	1.807638				
Panel D: Industry controls							
$\Delta No of Employees (1993-98)$	30,375	-1.620082	15.56758				
$\Delta Chinese_Import^{99 \rightarrow 07}$	30,375	2.470875	4.488955				
$\Delta Capital$ Intensity	30,375	.2019738	.1145263				
ΔICT Capital	30,375	.3862995	.1852666				
Initial Robot Density (1995)	30,375	4.39876	6.252753				

Table III: Exposure to Robots and Labor Market Outcomes

This table presents the coefficient estimates from the OLS and second-stage of the IV regressions for labor market outcomes. In all specifications, labor market outcome of interest is regressed on the changes in robot density between 1999 and 2007, initial observable household and contemporaneous industry characteristics and municipality dummies. In (1) and (2), we focus on the log changes in household earnings between 1999 and 2007, and in (3)-(4), the dependent variable is an indicator variable that takes the value of 1 if a household is unemployed in 2007, and zero otherwise. In (1) and (3), OLS regressions are estimated, while we estimate an IV regression in (2) and (4) instrumenting for the change in robot density in Sweden using the median change in robot density across the (non-Swedish) 11 European countries. Note that our base model, Equation 4, is defined and estimated in first differences, hence, our estimation model is equivalent to a fixed-effects regression. In all specifications, we account for potential spatial correlation across households within a municipality by clustering the standard errors at the municipality level. Statistical significance at the 10, 5, and 1 percent levels is indicated by *, **, and ***, respectively. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS, and OECD's STAN databases.

	Change in Earnings		Transition to Unemployment		
	OLS Estimates	IV Estimates	OLS Estimates	IV Estimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00859*	-0.01665***	0.00189***	0.00471***	
	(0.0048)	(0.0064)	(0.0006)	(0.0009)	
Age	0.01290	0.01278	-0.00103	-0.00098	
	(0.0127)	(0.0127)	(0.0020)	(0.0020)	
Age squared	-0.00053***	-0.00053***	0.00002	0.00002	
	(0.0002)	(0.0002)	(0.0000)	(0.0000)	
Male	0.34365^{***}	0.34334^{***}	-0.05067***	-0.05056***	
	(0.0384)	(0.0381)	(0.0051)	(0.0050)	
Married	0.10274^{***}	0.10243^{***}	-0.00749***	-0.00738***	
	(0.0176)	(0.0175)	(0.0026)	(0.0026)	
College and more	0.30450^{***}	0.30078^{***}	-0.02758^{***}	-0.02628***	
	(0.0291)	(0.0285)	(0.0039)	(0.0039)	
High school	0.13622^{***}	0.13436^{***}	-0.01090***	-0.01024^{***}	
	(0.0261)	(0.0260)	(0.0034)	(0.0033)	
Nbr of adults	0.12166^{***}	0.12198^{***}	-0.01361***	-0.01372^{***}	
	(0.0211)	(0.0209)	(0.0022)	(0.0022)	
Nbr of children	0.15651^{***}	0.15623^{***}	-0.00245^{*}	-0.00235^{*}	
	(0.0098)	(0.0097)	(0.0013)	(0.0013)	
Immigrant	-0.26335***	-0.26159^{***}	0.03275^{***}	0.03214^{***}	
	(0.0578)	(0.0572)	(0.0050)	(0.0050)	
$\Delta No of Employees (1993-98)$	-0.00041	0.00008	0.00022^{*}	0.00005	
	(0.0010)	(0.0010)	(0.0001)	(0.0001)	
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.00898**	-0.00962**	0.00097**	0.00119^{***}	
	(0.0039)	(0.0039)	(0.0005)	(0.0005)	
Δ Capital Intensity	-0.02452	-0.06565	-0.01709	-0.00267	
	(0.0865)	(0.0844)	(0.0125)	(0.0130)	
Δ ICT Capital	0.14936^{**}	0.13454^{**}	-0.00011	0.00509	
	(0.0637)	(0.0665)	(0.0086)	(0.0090)	
Initial Robot Density (1995)	0.00005	0.00208	-0.00109***	-0.00180***	
	(0.0027)	(0.0032)	(0.0004)	(0.0005)	
Constant	-0.20535	-0.16261	0.09676**	0.08178^{**}	
	(0.2316)	(0.2299)	(0.0374)	(0.0374)	
Observations	30,375	30,375	30,375	30,375	
R^2	0.0712	0.0711	0.0346	0.0337	
Clustering	Muni	Muni	Muni	Muni	
Municipality FEs	Yes	Yes	Yes	Yes	

Table IV: Exposure to Robots and Household Net Wealth

This table presents the coefficient estimates from the OLS and second-stage of the IV regressions for household net wealth. In all specifications, net wealth measures are regressed on the changes in robot density between 1999 and 2007, initial observable household and contemporaneous industry characteristics and municipality dummies. In (1) and (2), we focus on the percentile rank of the households in the within-cohort distribution, and in (3)-(4), the dependent variable is an indicator variable that takes the value of 1 if a household falls in the within-cohort net wealth distribution between 1999 and 2007, and 0 otherwise. In (1) and (3), OLS regressions are estimated, while we estimate an IV regression in (2) and (4) instrumenting for the change in robot density in Sweden using the median change in robot density across the (non-Swedish) 11 European countries. Note that our base model, Equation 4, is defined and estimated in first differences, hence, our estimation model is equivalent to a fixed-effects regression. In all specifications, we account for potential spatial correlation across households within a municipality by clustering the standard errors at the municipality level. Statistical significance at the 10, 5, and 1 percent levels is indicated by *, **, and ***, respectively. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS, and OECD's STAN databases.

	Net Wealth Rank (2007)		Downward Mobi	Downward Mobility (1999-2007)		
	OLS Estimates	IV Estimates	OLS Estimates	IVE stimates		
	(1)	(2)	(3)	(4)		
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.44696***	-0.78879***	0.00252**	0.00362**		
	(0.0763)	(0.0968)	(0.0012)	(0.0015)		
Age	-2.24288***	-2.24956^{***}	0.02977^{***}	0.02980***		
5	(0.2319)	(0.2289)	(0.0042)	(0.0042)		
Age squared	0.02483^{***}	0.02486^{***}	-0.00036***	-0.00036***		
	(0.0030)	(0.0030)	(0.0001)	(0.0001)		
Male	3.19852^{***}	3.18297^{***}	-0.03508***	-0.03503***		
	(0.4642)	(0.4621)	(0.0083)	(0.0082)		
Married	2.43364^{***}	2.41704^{***}	-0.00798	-0.00793		
	(0.4118)	(0.4087)	(0.0072)	(0.0071)		
College and more	12.94445^{***}	12.77618^{***}	-0.10699***	-0.10646***		
	(0.6240)	(0.6266)	(0.0097)	(0.0096)		
High school	4.70319***	4.62129^{***}	-0.02357^{***}	-0.02331***		
	(0.4196)	(0.4194)	(0.0077)	(0.0077)		
Nbr of adults	-1.09222^{***}	-1.09446***	-0.01299**	-0.01298**		
	(0.4168)	(0.4127)	(0.0064)	(0.0064)		
Nbr of children	-0.29157*	-0.30706*	0.00228	0.00233		
	(0.1653)	(0.1638)	(0.0029)	(0.0029)		
(IHS) Disposable Income	17.68274^{***}	17.72956^{***}	-0.11834^{***}	-0.11849***		
	(0.8389)	(0.8262)	(0.0114)	(0.0113)		
Immigrant	-11.32475^{***}	-11.24492^{***}	-0.00994	-0.01020		
	(0.5602)	(0.5563)	(0.0106)	(0.0106)		
$\Delta No \text{ of Employees (1993-98)}$	0.10119^{***}	0.12163^{***}	0.00020	0.00014		
	(0.0169)	(0.0173)	(0.0003)	(0.0003)		
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.14586^{***}	-0.17309^{***}	0.00324^{***}	0.00333^{***}		
	(0.0433)	(0.0437)	(0.0008)	(0.0008)		
Δ Capital Intensity	-2.88401	-4.63382**	0.15027^{***}	0.15587^{***}		
	(1.7537)	(1.8216)	(0.0306)	(0.0316)		
Δ ICT Capital	9.67114^{***}	9.04501***	-0.04804***	-0.04604**		
	(0.9941)	(0.9571)	(0.0184)	(0.0183)		
Initial Robot Density (1995)	-0.30682***	-0.22051***	-0.00038	-0.00066		
	(0.0539)	(0.0534)	(0.0007)	(0.0008)		
Constant	-138.30676^{***}	-137.04243^{***}	1.51765^{***}	1.51360^{***}		
	(10.3624)	(10.2412)	(0.1541)	(0.1532)		
Observations	30375	30375	30375	30375		
R^2	0.2333	0.2326	0.0643	0.0643		
Clustering	Muni	Muni	Muni	Muni		
Municipality FEs	Yes	Yes	Yes	Yes		

Table V: Exposure to Robots and Financial Risk Taking Behavior

This table presents the coefficient estimates from the OLS and second-stage of the IV regressions for household financial risk taking behavior. In all specifications, measures of financial risk taking are regressed on the changes in robot density between 1999 and 2007, initial observable household and contemporaneous industry characteristics and municipality dummies. In (1) and (2), we focus on the stockholding status in 2007, and in (3)-(4), the dependent variable is an indicator variable that takes the value of 1 if an incumbent stockholder household exits from the stock market as of 2007, and 0 otherwise. In (1) and (3), OLS regressions are estimated, while we estimate an IV regression in (2) and (4) instrumenting for the change in robot density in Sweden using the median change in robot density across the (non-Swedish) 11 European countries. Note that our base model, Equation 4, is defined and estimated in first differences, hence, our estimation model is equivalent to a fixed-effects regression. In all specifications, we account for potential spatial correlation across households within a municipality by clustering the standard errors at the municipality level. Statistical significance at the 10, 5, and 1 percent levels is indicated by *, **, and ***, respectively. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS, and OECD's STAN databases.

	Stockholdi	ng Status	Exit from the Stock Market		
	OLS Estimates	IV Estimates	OLS Estimates	IV Estimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot \ Density^{99 \rightarrow 07}$	-0.00308***	-0.00268*	0.00208**	0.00285**	
	(0.0010)	(0.0015)	(0.0009)	(0.0012)	
Age	-0.01011***	-0.01010***	0.00163	0.00166	
6	(0.0037)	(0.0037)	(0.0033)	(0.0033)	
Age squared	0.00004	0.00004	0.00001	0.00001	
	(0.0000)	(0.0000)	(0.0000)	(0.0000)	
Male	0.02366***	0.02368***	-0.01672***	-0.01671***	
	(0.0069)	(0.0068)	(0.0060)	(0.0060)	
Married	-0.01734***	-0.01732***	0.01242***	0.01244***	
	(0.0049)	(0.0049)	(0.0039)	(0.0038)	
College and more	0.13006***	0.13025***	-0.05850***	-0.05816***	
	(0.0084)	(0.0083)	(0.0066)	(0.0066)	
High school	0.06460***	0.06470***	-0.02446***	-0.02428***	
8	(0.0068)	(0.0067)	(0.0064)	(0.0063)	
Nbr of adults	-0.01320**	-0.01320**	0.00611	0.00613	
	(0.0057)	(0.0056)	(0.0045)	(0.0044)	
Nbr of children	0.00571**	0.00573**	-0.00407*	-0.00403*	
	(0.0026)	(0.0026)	(0.0023)	(0.0022)	
Wealth Quartile II	0.07014***	0.07016***	-0.03663***	-0.03662***	
foundin quantino m	(0.0070)	(0.0069)	(0.0064)	(0.0064)	
Wealth Quartile III	0.16863***	0.16867***	-0.07181***	-0.07179***	
foundin quantino III	(0.0066)	(0.0065)	(0.0058)	(0.0057)	
Wealth Quartile IV	0.21055***	0.21066***	-0.08855***	-0.08839***	
Would quartere IV	(0.0081)	(0.0081)	(0.0067)	(0.0066)	
(IHS) Disposable Income	0.23067***	0.23060***	-0.06308***	-0.06327***	
() P	(0.0082)	(0.0081)	(0.0068)	(0.0068)	
Immigrant	-0.13690***	-0.13698***	0.04824***	0.04809***	
	(0.0113)	(0.0112)	(0.0104)	(0.0103)	
ANo of Employees (1993-98)	0.00002	-0.00001	-0.00004	-0.00008	
((0.0002)	(0.0003)	(0.0002)	(0.0002)	
$\Delta Chinese \ Import^{99 \rightarrow 07}$	-0.00064	-0.00061	-0.00044	-0.00039	
port	(0.0006)	(0.0006)	(0.0005)	(0.0005)	
ACapital Intensity	0.06698***	0.06905***	-0.05477***	-0.05103***	
_ • • • F • • • • • • • • • • • • • • •	(0.0225)	(0.0226)	(0.0176)	(0.0178)	
AICT Capital	0.03484**	0.03557**	-0.00359	-0.00211	
	(0.0150)	(0.0152)	(0.0121)	(0.0123)	
Initial Robot Density (1995)	0.00086	0.00076	-0.00026	-0.00046	
	(0.0007)	(0.0008)	(0.0006)	(0.0006)	
Constant	-2.21683***	-2.21816***	1.00533***	1.00340***	
	(0.1214)	(0.1208)	(0.0982)	(0.0973)	
Observations	30375	30375	22125	22125	
R^2	0.1581	0.1581	0.0536	0.0536	
Clustering	Muni	Muni	Muni	Muni	
Municipality FEs	Yes	Yes	Yes	Yes	

Table VI: Exposure to Robots and Financial Wealth

This table presents the coefficient estimates from the OLS and second-stage of the IV regressions for household financial wealth accumulation. In all specifications, measures of financil wealth is regressed on the changes in robot density between 1999 and 2007, initial observable household and contemporaneous industry characteristics and municipality dummies. In (1) and (3), OLS regressions are estimated, while we estimate an IV regression in (2) and (4) instrumenting for the change in robot density in Sweden using the median change in robot density across the (non-Swedish) 11 European countries. In all specifications, we account for potential spatial correlation across households within a municipality by clustering the standard errors at the municipality level. Statistical significance at the 10, 5, and 1 percent levels is indicated by *, **, and ***, respectively. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS, and OECD's STAN databases.

	Changes in Fin	ancial Wealth	Wealth-to-Income Ratio		
	OLS Estimates	IV Estimates	OLS Estimates	IVE stimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.02758***	-0.04680***	-0.01695***	-0.02698***	
	(0.0075)	(0.0099)	(0.0045)	(0.0065)	
Age	-0.01347	-0.01381	-0.04789***	-0.04808***	
-	(0.0249)	(0.0247)	(0.0159)	(0.0158)	
Age squared	0.00023	0.00023	0.00097***	0.00097^{***}	
	(0.0003)	(0.0003)	(0.0002)	(0.0002)	
Male	0.37499^{***}	0.37415^{***}	-0.12373^{***}	-0.12405^{***}	
	(0.0514)	(0.0512)	(0.0346)	(0.0344)	
Married	0.12367^{***}	0.12274^{***}	-0.14464***	-0.14504***	
	(0.0326)	(0.0324)	(0.0254)	(0.0252)	
College and more	0.70377***	0.69486***	0.18282^{***}	0.17802^{***}	
C	(0.0476)	(0.0471)	(0.0371)	(0.0368)	
High school	0.29350***	0.28913^{***}	0.05408*	0.05175^{*}	
0	(0.0425)	(0.0422)	(0.0291)	(0.0289)	
Nbr of adults	-0.08304**	-0.08325**	-0.25239***	-0.25252***	
	(0.0381)	(0.0379)	(0.0352)	(0.0349)	
Nbr of children	-0.09889***	-0.09979***	-0.13548***	-0.13595***	
	(0.0169)	(0.0167)	(0.0119)	(0.0118)	
(IHS) Disposable Income	0.99564***	0.99939***	0.54586***	0.54749***	
, i,	(0.0496)	(0.0492)	(0.0815)	(0.0809)	
Immigrant	0.00356	0.00745	0.08762**	0.08967***	
0	(0.0603)	(0.0600)	(0.0338)	(0.0336)	
$\Delta No of Employees (1993-98)$	0.00300**	0.00415***	0.00453***	0.00513***	
F 5 Content of the second	(0.0015)	(0.0015)	(0.0013)	(0.0013)	
$\Delta Chinese \ Import^{99 \rightarrow 07}$	-0.01352***	-0.01505***	-0.00168	-0.00246	
	(0.0049)	(0.0048)	(0.0042)	(0.0043)	
Δ Capital Intensity	0.26109*	0.16290	-0.63089***	-0.68192***	
1 0	(0.1399)	(0.1377)	(0.1241)	(0.1243)	
Δ ICT Capital	0.30275***	0.26780***	0.28882***	0.27047***	
	(0.0965)	(0.0997)	(0.0818)	(0.0810)	
Initial Robot Density (1995)	-0.00601	-0.00116	-0.01187***	-0.00933***	
	(0.0044)	(0.0051)	(0.0026)	(0.0028)	
Financial Wealth in Logs (1999)	-0.66295***	-0.66324***	(,	(
	(0.0069)	(0.0068)			
(IHS) Net Wealth (1999)	(,		0.03138^{***}	0.03132^{***}	
			(0.0008)	(0.0008)	
Constant	-5.18899 * * *	-5.13122^{***}	-5.64021***	-5.60684***	
	(0.7767)	(0.7728)	(1.0282)	(1.0192)	
Observations	30375	30375	29955	29955	
R^2	0.5781	0.5780	0.0655	0.0652	
Clustering	Muni	Muni	Muni	Muni	
Municipality FEs	Yes	Yes	Yes	Yes	
Municipality FEs	res	res	res	res	

Table VII: Distributional Effects of Robots

This table presents the coefficient estimates from the second-stage of the IV regressions for various household economic variables. In (1) and (2), the dependent variable is a dummy variable that equals to 1 if a household, who was not saving for retirement in 1999, starts to contribute to the private retirement accounts in 2007. In (3) and (4), the dependent variable is a dummy variable that equals to 1 if a household, who was saving for retirement in 1999, stops contributing to the private retirement accounts in 2007. The dependent variable is regressed on the changes in robot density between 1999 and 2007, initial observable household and contemporaneous industry characteristics and municipality dummies. In (1) and (3), an OLS regression is estimated, while we estimate an IV regression in (2) and (4) instrumenting for the change in robot density in Sweden using the median change in robot density across the (non-Swedish) 11 European countries. Note that our base model, Equation 4, is defined and estimated in first differences, hence, our estimation model is equivalent to a fixed-effects regression. In all specifications, we account for potential spatial correlation across households within a municipality by clustering the standard errors at the municipality level. Statistical significance at the 10, 5, and 1 percent levels is indicated by *, **, and ***, respectively. Source: Author computations using household-level LINDA dataset from Statistics Sweden, data from the International Federation of Robotics (IFR) and the EU KLEMS, and OECD's STAN databases.

	Less- $Educated$	Better- $Educated$	Less- $Educated$	Better-Educated
	(1)	(2)	(3)	(4)
		Panel A. La	ıbor Market	
	Change i	n Earnings	Transition into	Unemployment
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.01727**	-0.01438*	0.00511***	0.00240*
	(0.0071)	(0.0083)	(0.0010)	(0.0014)
		Panel B. Hous	sehold Wealth	
	Percentile Ne	Percentile Net Wealth Rank		rd Mobility
$\Delta Robot_Density^{99 \rightarrow 07}$	-1.00243***	-0.25920	0.00456***	-0.00238
	(0.1083)	(0.1827)	(0.0017)	(0.0038)
	Panel C. Financial Risk Taking			
	Changes in Fi	inancial Wealth	Exit from the	e Stock Market
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.05355***	-0.01409	0.00429***	-0.00294
	(0.0115)	(0.0113)	(0.0014)	(0.0019)
II	\$7.	37	\$7.	X7
Household Controls	Yes	Yes	Yes	Yes
Industry Controls	Yes	Yes	Yes	Yes
Clustering	Muni	Muni	Muni	Muni
Municipality FEs	Yes	Yes	Yes	Yes

Appendix for Online Publication "Do Robots Increase Wealth Dispersion?"

THOMAS JANSSON and YIGITCAN KARABULUT

April 29, 2019

Abstract

This Online Appendix includes tables referred to but not included in the main body of the paper *Do Robots Increase Wealth Dispersion?* by Thomas Jansson and Yigitcan Karabulut, that provide robustness checks and additional findings.

SNI Code	EU-KLEMS Code	IFR Code	Industry Name (IFR)		
01-05	A-B	A-B	Agriculture, forestry, fishing		
\mathbf{C}	С	\mathbf{C}	Mining and quarrying		
15 -16	10-12	10-12	Food and beverages; tobacco		
17 - 18 - 19	13-15	13 - 15	Textiles		
20 - 21 - 22	16-18	16-18	Wood and furniture; Paper		
23-24	19-21	19-21	Pharmaceuticals, cosmetics; Other chemical products n.e.c.		
25 - 26	22-23	22 - 23	Rubber and plastic products; Chemical products; Mineral products		
27-28	24-25	24 - 25	Basic metals; Metal products (non-automotive)		
29	28	28	Industrial machinery		
30-31-32-33	26-27	26 - 27	Electrical/electronics		
34-35	29-30	29-30	Automotive; Other vehicles		
\mathbf{E}	${f E}$	\mathbf{E}	Electricity, gas, water supply		
\mathbf{F}	\mathbf{F}	\mathbf{F}	Construction		
Μ	Μ	Р	Education/research/development		

Table O.A.1: Mathcing the LINDA, IFR, and EU KLEMS Data

	Change in Income (incl. transfers)				
	OLS Estimates	IV Estimates			
	(1)	(2)			
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00268***	-0.00597***			
	(0.0008)	(0.0012)			
Age	-0.02660***	-0.02665^{***}			
	(0.0035)	(0.0034)			
Age squared	0.00024^{***}	0.00024^{***}			
	(0.0000)	(0.0000)			
Male	0.09214^{***}	0.09202^{***}			
	(0.0067)	(0.0066)			
Married	0.04945^{***}	0.04932^{***}			
	(0.0043)	(0.0043)			
College and more	0.07745^{***}	0.07593^{***}			
	(0.0065)	(0.0064)			
High school	0.01697^{***}	0.01621^{***}			
	(0.0049)	(0.0048)			
Nbr of adults	-0.10769^{***}	-0.10756^{***}			
	(0.0041)	(0.0041)			
Nbr of children	0.03484^{***}	0.03473^{***}			
	(0.0022)	(0.0022)			
Immigrant	0.01007	0.01078^{*}			
	(0.0066)	(0.0065)			
$\Delta No \text{ of Employees (1993-98)}$	-0.00008	0.00011			
	(0.0002)	(0.0002)			
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.00098*	-0.00125^{**}			
	(0.0005)	(0.0005)			
Δ Capital Intensity	-0.02844	-0.04523*			
	(0.0267)	(0.0270)			
Δ IT Capital	0.05255^{***}	0.04650***			
	(0.0135)	(0.0135)			
Initial Robot Density (1995)	-0.00103*	-0.00020			
	(0.0006)	(0.0006)			
Constant	1.12218^{***}	1.13963^{***}			
	(0.0655)	(0.0653)			
Observations	30375	30375			
R-squared	0.1011	0.1007			
Clustering	Muni	Muni			
Municipality FEs	Yes	Yes			

Table O.A.2: Exposure to Robots and Income Growth - Allowing for Transfers

	Upward Mobility (1999-2007)		
	OLS Estimates	IV Estimates	
	(1)	(2)	
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00293**	-0.00450***	
	(0.0012)	(0.0015)	
Age	-0.02770***	-0.02774^{***}	
	(0.0039)	(0.0039)	
Age squared	0.00039^{***}	0.00039^{***}	
	(0.0001)	(0.0001)	
Male	0.05088^{***}	0.05082^{***}	
	(0.0075)	(0.0075)	
Married	0.02148^{***}	0.02141^{***}	
	(0.0071)	(0.0071)	
College and more	0.14350^{***}	0.14278^{***}	
	(0.0094)	(0.0093)	
High school	0.04333^{***}	0.04298^{***}	
	(0.0071)	(0.0070)	
Nbr of adults	0.00578	0.00575	
	(0.0065)	(0.0065)	
Nbr of children	-0.00282	-0.00289	
	(0.0028)	(0.0028)	
(IHS) Disposable Income	0.16888^{***}	0.16916^{***}	
	(0.0102)	(0.0101)	
Immigrant	-0.05625***	-0.05594***	
-	(0.0093)	(0.0092)	
Net wealth quartile II (1999)	-0.28859***	-0.28870***	
-	(0.0133)	(0.0132)	
Net wealth quartile III (1999)	-0.39764^{***}	-0.39775***	
	(0.0163)	(0.0162)	
Net wealth quartile IV (1999)	-0.52697^{***}	-0.52739***	
-	(0.0133)	(0.0132)	
$\Delta No of Employees (1993-98)$	-0.00029	-0.00019	
	(0.0003)	(0.0003)	
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.00234^{***}	-0.00247***	
	(0.0007)	(0.0007)	
Δ Capital Intensity	-0.06369**	-0.07172^{**}	
	(0.0288)	(0.0295)	
ΔIT Capital	0.06563^{***}	0.06282^{***}	
	(0.0170)	(0.0169)	
Initial Robot Density (1995)	-0.00043	-0.00004	
	(0.0008)	(0.0008)	
Constant	-1.12314^{***}	-1.11815^{***}	
	(0.1412)	(0.1401)	
Observations	30375	30375	
R-squared	0.1813	0.1813	
Clustering	Muni	Muni	
Municipality FEs	Yes	Yes	

Table O.A.3: Exposure to Robots and Household NetWealth - Upward Mobility and Controlling for InitialWealth Quartiles

	Net Wealth H	Rank (2007)	Downward Mobility (1999-2007)	
	OLS Estimates	IVEstimates	OLS Estimates	IV Estimates
	(1)	(2)	(3)	(4)
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.39444***	-0.71109***	0.00254**	0.00365**
_ 0	(0.0703)	(0.0877)	(0.0012)	(0.0015)
Age	-2.75724^{***}	-2.76268***	0.02959***	0.02961^{***}
0	(0.2105)	(0.2079)	(0.0042)	(0.0042)
Age squared	0.03037^{***}	0.03039***	-0.00035***	-0.00035***
	(0.0027)	(0.0027)	(0.0001)	(0.0001)
Male	2.16855^{***}	2.15563^{***}	-0.03545***	-0.03541***
	(0.4388)	(0.4362)	(0.0083)	(0.0083)
Married	1.62278^{***}	1.60858^{***}	-0.00828	-0.00823
	(0.3606)	(0.3578)	(0.0073)	(0.0072)
College and more	12.02502***	11.87052***	-0.10733***	-0.10679***
5	(0.5281)	(0.5319)	(0.0097)	(0.0096)
High school	4.04676***	3.97187***	-0.02381***	-0.02355***
0	(0.4090)	(0.4096)	(0.0077)	(0.0076)
Nbr of adults	-1.74984***	-1.75096***	-0.01323**	-0.01322**
	(0.3712)	(0.3672)	(0.0064)	(0.0064)
Nbr of children	-0.91285***	-0.92629***	0.00205	0.00210
	(0.1396)	(0.1382)	(0.0029)	(0.0029)
(IHS) Disposable Income	12.89260***	12.94288***	-0.12009***	-0.12026***
	(0.7406)	(0.7293)	(0.0116)	(0.0116)
Immigrant	-7.77204***	-7.70324***	-0.00865	-0.00889
0	(0.5200)	(0.5174)	(0.0104)	(0.0104)
Homeowner	19.63302***	19.60465***	0.00715	0.00725
	(0.7222)	(0.7196)	(0.0108)	(0.0107)
$\Delta No of Employees (1993-98)$	0.10520***	0.12412***	0.00020	0.00014
r J i i i i i i i i i i i i i i i i i i	(0.0158)	(0.0161)	(0.0003)	(0.0003)
$\Delta Chinese \ Import^{99 \rightarrow 07}$	-0.14144***	-0.16667***	0.00325***	0.00333***
_ 1	(0.0405)	(0.0407)	(0.0008)	(0.0008)
$\Delta Capital Intensity$	-4.34611***	-5.96454***	0.14974***	0.15538^{***}
	(1.6294)	(1.6706)	(0.0307)	(0.0316)
Δ IT Capital	8.98883***	8.40994***	-0.04829***	-0.04627**
T T	(0.9972)	(0.9591)	(0.0184)	(0.0183)
Initial Robot Density (1995)	-0.30866***	-0.22872***	-0.00038	-0.00066
	(0.0473)	(0.0476)	(0.0007)	(0.0008)
Constant	-72.49908***	-71.42324***	1.54161***	1.53786***
	(9.4857)	(9.3807)	(0.1553)	(0.1543)
Observations	30375	30375	30375	30375
R-squared	0.3078	0.3072	0.0644	0.0643
Clustering	Muni	Muni	Muni	Muni
Municipality Fes	Yes	Yes	Yes	Yes

 Table O.A.4: Exposure to Robots and Household Net Wealth - Controlling for Homeownership

	Net Wealth Rank (2007)		Downward Mobi	lity (1999-2007)
	OLS Estimates	IV Estimates	OLS Estimates	IVE stimates
	(1)	(2)	(3)	(4)
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.44383***	-0.79439***	0.00254**	0.00359**
	(0.0752)	(0.0953)	(0.0012)	(0.0015)
Age	-2.21194^{***}	-2.21885^{***}	0.02995^{***}	0.02997^{***}
	(0.2313)	(0.2283)	(0.0042)	(0.0042)
Age squared	0.02460^{***}	0.02463^{***}	-0.00036***	-0.00036***
	(0.0030)	(0.0030)	(0.0001)	(0.0001)
Male	3.22733^{***}	3.21132^{***}	-0.03492***	-0.03487^{***}
	(0.4592)	(0.4574)	(0.0082)	(0.0082)
Married	2.52042^{***}	2.50322^{***}	-0.00750	-0.00745
	(0.4146)	(0.4115)	(0.0072)	(0.0071)
College and more	12.67147^{***}	12.49946^{***}	-0.10851***	-0.10800***
	(0.6096)	(0.6117)	(0.0097)	(0.0096)
High school	4.55422^{***}	4.47054^{***}	-0.02440***	-0.02415^{***}
	(0.4172)	(0.4168)	(0.0077)	(0.0077)
Nbr of adults	-1.07088^{**}	-1.07322^{***}	-0.01287^{**}	-0.01286**
	(0.4148)	(0.4106)	(0.0064)	(0.0063)
Nbr of children	-0.42370 **	-0.43932^{***}	0.00154	0.00159
	(0.1641)	(0.1624)	(0.0029)	(0.0029)
(IHS) Disposable Income	16.89054^{***}	16.94013^{***}	-0.12275^{***}	-0.12290^{***}
	(0.8032)	(0.7908)	(0.0113)	(0.0113)
Immigrant	-10.78501^{***}	-10.70422^{***}	-0.00694	-0.00718
	(0.5472)	(0.5429)	(0.0107)	(0.0106)
Risk exposure (1999)	4.11730^{***}	4.10907^{***}	0.02291^{***}	0.02293^{***}
	(0.4513)	(0.4502)	(0.0074)	(0.0073)
$\Delta No \text{ of Employees (1993-98)}$	0.10277^{***}	0.12372^{***}	0.00021	0.00015
	(0.0167)	(0.0171)	(0.0003)	(0.0003)
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.15279^{***}	-0.18070^{***}	0.00320***	0.00329^{***}
	(0.0424)	(0.0427)	(0.0008)	(0.0008)
Δ Capital Intensity	-3.33717*	-5.13070^{***}	0.14775^{***}	0.15310^{***}
	(1.7415)	(1.8053)	(0.0305)	(0.0315)
Δ IT Capital	9.70795^{***}	9.06578^{***}	-0.04784***	-0.04592^{**}
	(0.9983)	(0.9604)	(0.0184)	(0.0183)
Initial Robot Density (1995)	-0.30949***	-0.22097^{***}	-0.00039	-0.00066
	(0.0532)	(0.0527)	(0.0007)	(0.0008)
Constant	-130.60356^{***}	-129.32241^{***}	1.56051^{***}	1.55668^{***}
	(10.0987)	(9.9737)	(0.1537)	(0.1528)
Observations	30375	30375	30375	30375
R-squared	0.2369	0.2361	0.0647	0.0646
Clustering	Muni	Muni	Muni	Muni
Municipality Fes	Yes	Yes	Yes	Yes

Table O.A.5: Exposure to Robots and Household Net Wealth - Controlling for Risk Exposure

	Char	nges in Househo	old Debt (1999-20	07)
	OLS Estimates	IV Estimates	OLS Estimates	IV Estimates
	(1)	(2)	(3)	(4)
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00191	-0.00191	-0.00471	-0.00471
	(0.0124)	(0.0105)	(0.0092)	(0.0085)
Age	-0.42759***	-0.42759***	-0.21089***	-0.21089***
5	(0.0409)	(0.0390)	(0.0312)	(0.0291)
Age squared	0.00431^{***}	0.00431^{***}	0.00155^{***}	0.00155^{***}
	(0.0005)	(0.0005)	(0.0004)	(0.0004)
Male	0.12304	0.12304	0.34546^{***}	0.34546^{***}
	(0.0949)	(0.0757)	(0.0739)	(0.0609)
Married	-0.10868**	-0.10868**	0.13364^{***}	0.13364^{***}
	(0.0511)	(0.0530)	(0.0406)	(0.0400)
College and more	0.47686^{***}	0.47686^{***}	0.24851^{***}	0.24851^{***}
-	(0.0808)	(0.0740)	(0.0522)	(0.0560)
High school	0.04155	0.04155	0.10292^{**}	0.10292^{**}
	(0.0575)	(0.0541)	(0.0500)	(0.0458)
Nbr of adults	0.03720	0.03720	0.29704^{***}	0.29704^{***}
	(0.0522)	(0.0503)	(0.0404)	(0.0401)
Nbr of children	-0.03084	-0.03084	0.21485^{***}	0.21485^{***}
	(0.0244)	(0.0225)	(0.0194)	(0.0184)
(IHS) Disposable Income	-0.72904^{***}	-0.72904^{***}	0.61667^{***}	0.61667^{***}
	(0.0998)	(0.0998)	(0.1086)	(0.0850)
Immigrant	0.37300^{***}	0.37300^{***}	-0.19504^{***}	-0.19504***
	(0.0804)	(0.0915)	(0.0646)	(0.0740)
$\Delta No of Employees (1993-98)$	-0.00286	-0.00286	-0.00268	-0.00268
	(0.0029)	(0.0022)	(0.0020)	(0.0018)
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.00410	-0.00410	-0.00442	-0.00442
	(0.0072)	(0.0060)	(0.0051)	(0.0046)
Δ Capital Intensity	-0.53097**	-0.53097**	-0.18900	-0.18900
	(0.2422)	(0.2230)	(0.1855)	(0.1739)
Δ IT Capital	0.24082	0.24082	0.18206	0.18206
	(0.1495)	(0.1555)	(0.1303)	(0.1240)
Initial Robot Density (1995)	0.00047	0.00047	0.00106	0.00106
	(0.0061)	(0.0055)	(0.0056)	(0.0048)
Household Debt in Logs (1999)			-0.60806***	-0.60806***
_			(0.0108)	(0.0091)
Constant	20.35497^{***}	20.35497^{***}	3.89887^{***}	3.89887***
	(1.5248)	(1.4380)	(1.4419)	(1.2171)
Observations	30375	30375	30375	30375
R-squared	0.0759	0.0759	0.4049	0.4049
Clustering	Muni	Muni	Muni	Muni
Municipality Fes	Yes	Yes	Yes	Yes

Table O.A.6: Exposure to Robots and Changes in Household Debt

	Net Wealth	Rank (2007)	Downward Mobility (1999-2007)	
	OLS Estimates IV Estimates		OLS Estimates	IV Estimates
	(1)	(2)	(3)	(4)
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.43786***	-0.77112***	0.00244**	0.00345**
_ v	(0.0755)	(0.0964)	(0.0012)	(0.0015)
Age	-2.25541^{***}	-2.26185***	0.02989***	0.02991***
0	(0.2332)	(0.2302)	(0.0042)	(0.0042)
Age squared	0.02537^{***}	0.02540^{***}	-0.00036***	-0.00036***
	(0.0030)	(0.0030)	(0.0001)	(0.0001)
Male	2.83914^{***}	2.82610^{***}	-0.03180***	-0.03176***
	(0.4649)	(0.4629)	(0.0083)	(0.0082)
Married	2.32813^{***}	2.31258^{***}	-0.00702	-0.00697
	(0.4114)	(0.4085)	(0.0072)	(0.0072)
College and more	12.63206^{***}	12.46989^{***}	-0.10414***	-0.10365^{***}
-	(0.6214)	(0.6243)	(0.0097)	(0.0097)
High school	4.56213^{***}	4.48314^{***}	-0.02229***	-0.02204***
-	(0.4169)	(0.4167)	(0.0077)	(0.0077)
Nbr of adults	-1.20851^{***}	-1.21001^{***}	-0.01193*	-0.01192*
	(0.4143)	(0.4102)	(0.0064)	(0.0064)
Nbr of children	-0.45333^{***}	-0.46747^{***}	0.00375	0.00380
	(0.1670)	(0.1655)	(0.0029)	(0.0029)
(IHS) Disposable Income	17.64789***	17.69373^{***}	-0.11802***	-0.11816***
-	(0.8230)	(0.8103)	(0.0114)	(0.0113)
Immigrant	-11.05198 ***	-10.97578^{***}	-0.01244	-0.01267
-	(0.5605)	(0.5574)	(0.0105)	(0.0104)
Income Growth (1999 - 2007)	1.05096^{***}	1.04476^{***}	-0.00960***	-0.00958***
	(0.0893)	(0.0887)	(0.0020)	(0.0020)
$\Delta No of Employees (1993-98)$	0.10169^{***}	0.12160^{***}	0.00020	0.00014
	(0.0168)	(0.0172)	(0.0003)	(0.0003)
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.13646^{***}	-0.16306***	0.00316^{***}	0.00324^{***}
-	(0.0424)	(0.0429)	(0.0008)	(0.0008)
Δ Capital Intensity	-2.85370	-4.55955**	0.14999^{***}	0.15519^{***}
	(1.7362)	(1.8041)	(0.0306)	(0.0316)
Δ IT Capital	9.51263^{***}	8.90324^{***}	-0.04660**	-0.04474**
-	(0.9782)	(0.9425)	(0.0184)	(0.0184)
Initial Robot Density (1995)	-0.30692***	-0.22279***	-0.00038	-0.00064
-	(0.0529)	(0.0523)	(0.0007)	(0.0008)
Constant	-137.68344^{***}	-136.45469^{***}	1.51196^{***}	1.50821^{***}
	(10.1443)	(10.0303)	(0.1537)	(0.1528)
Observations	30375	30375	30375	30375
R-squared	0.2366	0.2359	0.0652	0.0651
Clustering	Muni	Muni	Muni	Muni
Municipality Fes	Yes	Yes	Yes	Yes

Table O.A.7: Exposure to Robots and Household Net Wealth - Controlling for Income Growth

	Net Wealth H	Rank (2007)	Downward Mobility (1999-2007)		
	OLS Estimates	IVEstimates	OLS Estimates	IV Estimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.39185***	-0.69194***	0.00510***	0.00700***	
_ 0	(0.0746)	(0.0960)	(0.0014)	(0.0017)	
Age	-2.09862***	-2.11012***	0.02265***	0.02272***	
5	(0.2381)	(0.2353)	(0.0046)	(0.0046)	
Age squared	0.02115^{***}	0.02124^{***}	-0.00033***	-0.00033***	
0	(0.0030)	(0.0030)	(0.0001)	(0.0001)	
Male	2.64276^{***}	2.63037^{***}	-0.03268***	-0.03260***	
	(0.4720)	(0.4685)	(0.0091)	(0.0090)	
Married	1.79308***	1.78724***	-0.01324*	-0.01321*	
	(0.3703)	(0.3662)	(0.0076)	(0.0075)	
College and more	10.43474***	10.29303***	-0.10816***	-0.10726***	
5	(0.5577)	(0.5565)	(0.0101)	(0.0100)	
High school	3.67313***	3.59642***	-0.03267***	-0.03219***	
0	(0.4782)	(0.4763)	(0.0087)	(0.0086)	
Nbr of adults	-0.07494	-0.08374	-0.00781	-0.00776	
	(0.3509)	(0.3470)	(0.0073)	(0.0072)	
Nbr of children	0.12044	0.10559	0.00178	0.00187	
	(0.1646)	(0.1640)	(0.0035)	(0.0035)	
(IHS) Disposable Income	13.65140***	13.72410***	-0.10322***	-0.10368***	
, , , , , , , , , , , , , , , , , , ,	(0.7109)	(0.7042)	(0.0130)	(0.0128)	
Immigrant	-6.88723***	-6.83765***	0.00979	0.00948	
0	(0.5871)	(0.5857)	(0.0132)	(0.0131)	
Total saving rate (2000)	2.19225^{***}	2.20285***	-0.08094***	-0.08100***	
	(0.2685)	(0.2647)	(0.0048)	(0.0047)	
ΔN_0 of Employees (1993-98)	0.10162***	0.11784***	-0.00088**	-0.00098***	
FJ ()	(0.0165)	(0.0167)	(0.0003)	(0.0004)	
$\Delta Chinese \ Import^{99 \rightarrow 07}$	-0.17900***	-0.19992***	0.00325***	0.00338***	
_ 1	(0.0440)	(0.0439)	(0.0010)	(0.0010)	
$\Delta Capital Intensity$	-10.15168***	-11.50817***	0.19317***	0.20174^{***}	
	(1.7642)	(1.8045)	(0.0310)	(0.0319)	
Δ IT Capital	9.76399***	9.16267***	-0.08074***	-0.07694***	
T. T. T.	(1.1425)	(1.1031)	(0.0231)	(0.0229)	
Robot Density (1995)	-0.31852***	-0.23940***	0.00198**	0.00148*	
	(0.0473)	(0.0467)	(0.0009)	(0.0008)	
Constant	-74.53264***	-73.75948***	1.59149***	1.58661***	
	(9.3058)	(9.2042)	(0.1735)	(0.1721)	
Observations	20933	20933	20933	20933	
R-squared	0.2374	0.2366	0.1340	0.1339	
Clustering	Muni	Muni	Muni	Muni	
Municipality FEs	Yes	Yes	Yes	Yes	

Table O.A.8: Exposure to Robots and Household Net Wealth - Controlling for Total Saving Rate in 2000 (Normalized by Current Income)

	Net Wealth H	Rank (2007)	Downward Mobi	Downward Mobility (1999-2007	
	OLS Estimates IV Estim		OLS Estimates	IV Estimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.37517***	-0.66804***	0.00483***	0.00671***	
_ 0	(0.0737)	(0.0958)	(0.0014)	(0.0018)	
Age	-1.92988***	-1.94054***	0.01980***	0.01987***	
0	(0.2280)	(0.2263)	(0.0049)	(0.0048)	
Age squared	0.01931^{***}	0.01939^{***}	-0.00030***	-0.00030***	
	(0.0029)	(0.0028)	(0.0001)	(0.0001)	
Male	2.66917^{***}	2.65710^{***}	-0.03225***	-0.03217^{***}	
	(0.4601)	(0.4574)	(0.0101)	(0.0100)	
Married	1.92899^{***}	1.92371^{***}	-0.01524*	-0.01521*	
	(0.3579)	(0.3549)	(0.0078)	(0.0078)	
College and more	10.46690***	10.32903^{***}	-0.11366***	-0.11277^{***}	
5	(0.5349)	(0.5321)	(0.0111)	(0.0111)	
High school	3.65475^{***}	3.57977^{***}	-0.03190***	-0.03142^{***}	
C	(0.4271)	(0.4250)	(0.0088)	(0.0087)	
Nbr of adults	-0.14505	-0.15380	-0.00768	-0.00762	
	(0.3458)	(0.3432)	(0.0074)	(0.0074)	
Nbr of children	0.16129	0.14697	0.00056	0.00066	
	(0.1628)	(0.1615)	(0.0033)	(0.0033)	
(IHS) Disposable Income	13.39191^{***}	13.46160^{***}	-0.09294***	-0.09339***	
-	(0.6544)	(0.6473)	(0.0130)	(0.0129)	
Immigrant	-7.13208***	-7.08506***	0.02204^{*}	0.02174^{*}	
C	(0.5724)	(0.5685)	(0.0126)	(0.0126)	
Total saving rate (2000)	-0.25830**	-0.25519**	-0.05262***	-0.05264***	
<u> </u>	(0.1151)	(0.1143)	(0.0023)	(0.0022)	
Δ No of Employees (1993-98)	0.09338^{***}	0.10918^{***}	-0.00078**	-0.00088***	
	(0.0173)	(0.0180)	(0.0003)	(0.0003)	
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.18195***	-0.20239***	0.00344^{***}	0.00357^{***}	
-	(0.0438)	(0.0440)	(0.0010)	(0.0010)	
Δ Capital Intensity	-9.30398***	-10.62448^{***}	0.16717^{***}	0.17566^{***}	
	(1.8112)	(1.8553)	(0.0330)	(0.0336)	
Δ IT Capital	9.63583***	9.04853***	-0.08014***	-0.07636***	
_	(1.1653)	(1.1468)	(0.0211)	(0.0208)	
Robot Density (1995)	-0.30712^{***}	-0.22986***	0.00197**	0.00147^{*}	
	(0.0445)	(0.0447)	(0.0008)	(0.0009)	
Constant	-74.07100***	-73.31138^{***}	1.53527^{***}	1.53039^{***}	
	(10.4745)	(10.3418)	(0.1923)	(0.1907)	
Observations	20933	20933	20933	20933	
R-squared	0.2299	0.2292	0.1366	0.1366	
Clustering	Muni	Muni	Muni	Muni	
Municipality Fes	Yes	Yes	Yes	Yes	

Table O.A.9: Exposure to Robots and Household Net Wealth - Controlling for Total Saving Rate in 2000 (Normalized by Wealth from Prior Period)

	Net Wealth	Rank (2007)	Downward Mobility (1999-2007	
	OLS Estimates	IV Estimates	OLS Estimates	IV Estimates
	(1)	(2)	(3)	(4)
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.44611***	-0.77606***	0.00251**	0.00349**
_ •	(0.0763)	(0.0962)	(0.0012)	(0.0015)
Age	-2.24590***	-2.25233***	0.02980***	0.02982***
5	(0.2303)	(0.2274)	(0.0042)	(0.0042)
Age squared	0.02459^{***}	0.02462^{***}	-0.00035***	-0.00035***
	(0.0030)	(0.0030)	(0.0001)	(0.0001)
Male	3.20539^{***}	3.19040***	-0.03515***	-0.03511***
	(0.4607)	(0.4585)	(0.0082)	(0.0081)
Married	2.34529***	2.32930***	-0.00708	-0.00703
	(0.4077)	(0.4047)	(0.0072)	(0.0071)
College and more	12.60889***	12.44656***	-0.10356***	-0.10309***
	(0.6089)	(0.6121)	(0.0096)	(0.0095)
High school	4.59262***	4.51364***	-0.02245***	-0.02222***
0	(0.4127)	(0.4125)	(0.0076)	(0.0076)
Nbr of adults	-0.71337*	-0.71554*	-0.01688**	-0.01687***
	(0.4188)	(0.4149)	(0.0065)	(0.0065)
Nbr of children	-0.30064*	-0.31561**	0.00237	0.00242
	(0.1603)	(0.1590)	(0.0029)	(0.0028)
(IHS) Disposable Income	17.37567***	17.42088***	-0.11519***	-0.11533***
	(0.8234)	(0.8114)	(0.0115)	(0.0115)
Immigrant	-11.11618***	-11.03898***	-0.01211	-0.01234
0	(0.5596)	(0.5554)	(0.0107)	(0.0106)
Active saving rate (2000)	9.59107***	9.58937***	-0.09820***	-0.09819***
8	(0.5015)	(0.4982)	(0.0085)	(0.0084)
ΔNo of Employees (1993-98)	0.10187***	0.12160***	0.00019	0.00014
F5	(0.0167)	(0.0171)	(0.0003)	(0.0003)
$\Delta Chinese \ Import^{99 \rightarrow 07}$	-0.13137***	-0.15766***	0.00310***	0.00317***
_ 1	(0.0430)	(0.0435)	(0.0008)	(0.0008)
$\Delta Capital Intensity$	-2.30459	-3.99368**	0.14434^{***}	0.14932***
I IIIIIIII	(1.7525)	(1.8198)	(0.0306)	(0.0316)
Δ IT Capital	9.25158***	8.64703***	-0.04370**	-0.04192**
I III	(0.9888)	(0.9517)	(0.0184)	(0.0184)
Robot Density (1995)	-0.29434***	-0.21104***	-0.00051	-0.00075
	(0.0544)	(0.0537)	(0.0007)	(0.0008)
Constant	-134.66414***	-133.44434***	1.48034***	1.47674***
	(10.1513)	(10.0406)	(0.1538)	(0.1528)
Observations	30374	30374	30374	30374
R-squared	0.2469	0.2462	0.0685	0.0685
Clustering	Muni	Muni	Muni	Muni
M	Vog	Vog	Voq	Vog

Table O.A.10: Exposure to Robots and Household Net Wealth - Controlling for ActiveSaving Rate in 2000

	Net Wealth F	Rank (2007)	Downward Mobi	lity (1999-2007)
	OLS Estimates	IVEstimates	OLS Estimates	IV Estimates
	(1)	(2)	(3)	(4)
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.20653***	-0.32478***	0.00449***	0.00708***
	(0.0496)	(0.0636)	(0.0012)	(0.0015)
Age	-2.22034^{***}	-2.22294^{***}	0.02851^{***}	0.02857^{***}
	(0.1782)	(0.1767)	(0.0040)	(0.0040)
Age squared	0.01732^{***}	0.01733^{***}	-0.00040***	-0.00040***
	(0.0023)	(0.0023)	(0.0001)	(0.0001)
Male	1.93656^{***}	1.93167^{***}	-0.04598***	-0.04587^{***}
	(0.3568)	(0.3540)	(0.0077)	(0.0077)
Married	0.79514^{***}	0.79012^{***}	-0.02249***	-0.02238***
	(0.2726)	(0.2709)	(0.0071)	(0.0071)
College and more	7.25287^{***}	7.19816***	-0.14675^{***}	-0.14556^{***}
5	(0.4485)	(0.4454)	(0.0097)	(0.0097)
High school	2.34184^{***}	2.31489^{***}	-0.04419***	-0.04360***
2	(0.3067)	(0.3049)	(0.0072)	(0.0072)
Nbr of adults	-0.19703	-0.19863	-0.00417	-0.00414
	(0.2417)	(0.2395)	(0.0064)	(0.0063)
Nbr of children	-0.32738***	-0.33272***	0.00191	0.00203
	(0.1088)	(0.1078)	(0.0028)	(0.0028)
(IHS) Disposable Income	9.09334***	9.11473***	-0.17855***	-0.17902***
	(0.5211)	(0.5155)	(0.0119)	(0.0118)
Immigrant	-4.37404***	-4.35017***	0.06035***	0.05983***
C	(0.3862)	(0.3831)	(0.0100)	(0.0099)
Wealth Quartile II (1999)	13.23806***	13.23006***	0.29230***	0.29248***
•	(0.3355)	(0.3329)	(0.0133)	(0.0132)
Wealth Quartile III (1999)	29.60435***	29.59584***	0.39496***	0.39514^{***}
•	(0.4061)	(0.4028)	(0.0167)	(0.0166)
Wealth Quartile IV (1999)	50.08486***	50.05290***	0.45317***	0.45387***
•	(0.5743)	(0.5677)	(0.0133)	(0.0132)
$\Delta No of Employees (1993-98)$	0.02769**	0.03481^{***}	-0.00027	-0.00042
	(0.0119)	(0.0118)	(0.0003)	(0.0003)
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.12677***	-0.13620***	0.00299***	0.00320***
- 1	(0.0297)	(0.0296)	(0.0007)	(0.0008)
Δ Capital Intensity	-1.84733	-2.45467**	0.12669^{***}	0.13999***
1 0	(1.1811)	(1.2264)	(0.0291)	(0.0299)
∆IT Capital	4.18632***	3.97380***	-0.08529***	-0.08063***
L	(0.7741)	(0.7640)	(0.0173)	(0.0171)
Initial Robot Density (1995)	-0.08092**	-0.05126	0.00116	0.00051
	(0.0326)	(0.0355)	(0.0008)	(0.0008)
Constant	-32.16498***	-31.78803***	2.18566***	2.17741***
	(6.7957)	(6.7368)	(0.1569)	(0.1556)
Observations	30375	30375	30375	30375
R-squared	0.5624	0.5623	0.1693	0.1692
Clustering	Muni	Muni	Muni	Muni
Municipality Fes	Yes	Yes	Yes	Yes

 Table O.A.11: Exposure to Robots and Household Net Wealth - Controlling for Initial

 Wealth Quartiles

	Net Wealth I	Rank (2007)	Downward Mobility (1999-2007)		
	OLS Estimates IV Estimates		OLS Estimates	IV Estimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.28598***	-0.53625***	0.00452***	0.00676***	
-	(0.0590)	(0.0760)	(0.0012)	(0.0015)	
Age	-2.43016^{***}	-2.43482***	0.02745^{***}	0.02749^{***}	
0	(0.1901)	(0.1882)	(0.0040)	(0.0040)	
Age squared	0.02302***	0.02304^{***}	-0.00038***	-0.00038***	
0	(0.0025)	(0.0024)	(0.0001)	(0.0001)	
Male	2.30494^{***}	2.29462***	-0.04617***	-0.04608***	
	(0.3641)	(0.3625)	(0.0078)	(0.0077)	
Married	1.52273^{***}	1.51167^{***}	-0.01929***	-0.01919***	
	(0.3227)	(0.3206)	(0.0071)	(0.0071)	
College and more	10.23673^{***}	10.11684^{***}	-0.14061***	-0.13954^{***}	
C	(0.4670)	(0.4668)	(0.0095)	(0.0094)	
High school	3.00960***	2.95170***	-0.04460***	-0.04408***	
C	(0.3458)	(0.3448)	(0.0071)	(0.0071)	
Nbr of adults	-0.44063	-0.44304	-0.00490	-0.00488	
	(0.3073)	(0.3040)	(0.0062)	(0.0062)	
Nbr of children	-0.34475***	-0.35602***	0.00162	0.00172	
	(0.1275)	(0.1266)	(0.0028)	(0.0028)	
(IHS) Disposable Income	13.10548^{***}	13.14517^{***}	-0.17517***	-0.17553***	
· · · •	(0.6275)	(0.6191)	(0.0112)	(0.0111)	
Immigrant	-5.78479***	-5.73296***	0.05884^{***}	0.05837^{***}	
-	(0.3957)	(0.3939)	(0.0104)	(0.0103)	
(IHS) Net Wealth (1999)	1.15618^{***}	1.15481^{***}	0.01435^{***}	0.01437^{***}	
	(0.0153)	(0.0152)	(0.0005)	(0.0005)	
Δ No of Employees (1993-98)	0.07358^{***}	0.08856^{***}	-0.00014	-0.00028	
	(0.0137)	(0.0138)	(0.0003)	(0.0003)	
$\Delta Chinese_Import^{99 \rightarrow 07}$	-0.16532^{***}	-0.18522^{***}	0.00300***	0.00318^{***}	
	(0.0340)	(0.0340)	(0.0008)	(0.0008)	
Δ Capital Intensity	-5.38992***	-6.66709***	0.11916^{***}	0.13058^{***}	
	(1.4106)	(1.4478)	(0.0285)	(0.0295)	
∆IT Capital	7.24189^{***}	6.78670***	-0.07820***	-0.07413^{***}	
	(0.9029)	(0.8805)	(0.0174)	(0.0172)	
Initial Robot Density (1995)	-0.22577 ***	-0.16272^{***}	0.00063	0.00006	
	(0.0390)	(0.0399)	(0.0008)	(0.0008)	
Constant	-72.48685^{***}	-71.63999 ***	2.33484^{***}	2.32727^{***}	
	(8.2177)	(8.1287)	(0.1463)	(0.1452)	
Observations	30375	30375	30375	30375	
R-squared	0.4529	0.4525	0.1646	0.1645	
Clustering	Muni	Muni	Muni	Muni	
Municipality Fes	Yes	Yes	Yes	Yes	

Table O.A.12: Exposure to Robots and Household Net Wealth - Controlling for Initial Wealth in Levels

	Net Wealth Rank (2007)		Downward Mobility (1999-2007)		
	OLS Estimates	IV Estimates	OLS Estimates	IV Estimates	
	(1)	(2)	(3)	(4)	
$\Delta Robot_Density^{99\rightarrow07}$	-0.40893***	-0.68396***	0.00217*	0.00263*	
	(0.0709)	(0.0876)	(0.0011)	(0.0014)	
Age	-2.41885***	-2.42410***	0.03142^{***}	0.03143^{***}	
5	(0.2179)	(0.2154)	(0.0042)	(0.0042)	
Age squared	0.02597^{***}	0.02599***	-0.00037***	-0.00037***	
	(0.0028)	(0.0028)	(0.0001)	(0.0001)	
Male	2.71370^{***}	2.70153^{***}	-0.03054***	-0.03052***	
	(0.4450)	(0.4422)	(0.0080)	(0.0079)	
Married	2.46785^{***}	2.45447^{***}	-0.00830	-0.00828	
	(0.3886)	(0.3858)	(0.0072)	(0.0071)	
College and more	11.04960***	10.91560***	-0.08925***	-0.08903***	
5	(0.6161)	(0.6172)	(0.0094)	(0.0094)	
High school	4.10072^{***}	4.03527***	-0.01793**	-0.01782**	
5	(0.4054)	(0.4042)	(0.0076)	(0.0075)	
Nbr of adults	0.00373	0.00113	-0.02325***	-0.02324***	
	(0.4141)	(0.4103)	(0.0066)	(0.0065)	
Nbr of children	0.17930	0.16650	-0.00213	-0.00211	
	(0.1634)	(0.1624)	(0.0028)	(0.0028)	
(IHS) Disposable Income	16.06876***	16.10759***	-0.10323***	-0.10330***	
	(0.8453)	(0.8340)	(0.0115)	(0.0115)	
Immigrant	-11.04372^{***}	-10.97970***	-0.01258	-0.01268	
C	(0.5264)	(0.5233)	(0.0107)	(0.0106)	
Mean active saving rate (2000-2007)	77.44070***	77.38476***	-0.72506***	-0.72496***	
<u> </u>	(2.3000)	(2.2798)	(0.0320)	(0.0318)	
$\Delta No of Employees (1993-98)$	0.08837***	0.10482***	0.00032	0.00029	
	(0.0155)	(0.0158)	(0.0003)	(0.0003)	
$\Delta Chinese \ Import^{99 \rightarrow 07}$	-0.07775*	-0.09971**	0.00261***	0.00264***	
_ 1	(0.0406)	(0.0408)	(0.0008)	(0.0008)	
Δ Capital Intensity	-0.85246	-2.26166	0.13125***	0.13365***	
i v	(1.6539)	(1.7098)	(0.0305)	(0.0315)	
Δ IT Capital	7.32673***	6.82470***	-0.02609	-0.02524	
L	(0.9438)	(0.9107)	(0.0189)	(0.0188)	
Initial Robot Density (1995)	-0.25097***	-0.18157***	-0.00090	-0.00102	
	(0.0473)	(0.0465)	(0.0007)	(0.0008)	
Constant	-117.29645***	-116.29447***	1.32094***	1.31923***	
	(10.1761)	(10.0629)	(0.1529)	(0.1519)	
Observations	30375	30375	30375	30375	
R-squared	0.3270	0.3265	0.0887	0.0887	
Clustering	Muni	Muni	Muni	Muni	
Municipality FEs	Yes	Yes	Yes	Yes	

Table O.A.13: Exposure to Robots and Household Net Wealth - Controlling for Average Active Saving Rate between 1999-2007

	OLS Estimates	IVEstimates	OLS Estimates	IVE stimates
	(1)	(2)	(3)	(4)
		Panel A. La	abor Market	
	Change in	Earnings	Transition into I	Unemployment
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00950*	-0.02148***	0.00175***	0.00426***
	(0.0049)	(0.0064)	(0.0007)	(0.0009)
		Panel B. Hou	sehold Wealth	
	Percentile Net	Wealth Rank	Downward	l Mobility
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.48822***	-0.97492***	0.00247**	0.00369**
	(0.0755)	(0.0908)	(0.0012)	(0.0015)
	Р	anel C. Finan	cial Risk Taking	5
	Exit from the S	Stock Market	Wealth-to-In	come Ratio
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00244***	0.00406***	-0.02145***	-0.03651***
	(0.0009)	(0.0011)	(0.0048)	(0.0068)
			77	37
Household Controls	Yes	Yes	Yes	Yes
Industry Controls	Yes	Yes	Yes	Yes
Clustering	Muni	Muni	Muni	Muni
Municipality FEs	Yes	Yes	Yes	Yes

Table O.A.14: Exposure to Robots and Household Economic Behavior - Excluding the Automative Industry

	OLS Estimates	IVEstimates	OLS Estimates	IVE stimates
	(1)	(2)	(3)	(4)
		Panel A. La	abor Market	
	Change in	Earnings	Transition into I	Unemployment
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.01875***	-0.02576***	0.00438***	0.00755***
	(0.0072)	(0.0088)	(0.0010)	(0.0013)
		Panel B. Hou	sehold Wealth	
	Percentile Net	Wealth Rank	Downward	Mobility
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.79471***	-1.14770***	0.00257	0.00421*
	(0.1083)	(0.1373)	(0.0018)	(0.0023)
	Р	anel C. Finan	cial Risk Taking	5
	Exit from the S	Stock Market	Wealth-to-In	come Ratio
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00434***	0.00444**	-0.02345***	-0.04220***
	(0.0014)	(0.0018)	(0.0078)	(0.0102)
Hereicheld Controls	X 7	X 7	V	V.
Household Controls	res	res	res	res
Industry Controls	Yes	Yes	Yes	Yes
Clustering	Muni	Muni	Muni	Muni
Municipality FEs	Yes	Yes	Yes	Yes

Table O.A.15: Exposure to Robots and Household Economic Behavior - Excluding the Rubber and Plastic Industry

	OLS Estimates	IVEstimates	OLS Estimates	IVE stimates
	(1)	(2)	(3)	(4)
		Panel A. La	abor Market	
	Change in	Earnings	Transition into I	Unemployment
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00778	-0.01796*	0.00145*	0.00519***
	(0.0071)	(0.0098)	(0.0008)	(0.0010)
		Panel B. Hou	sehold Wealth	
	Percentile Net	Wealth Rank	Downward	l Mobility
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.37680***	-0.69265***	0.00274*	0.00415**
	(0.0850)	(0.1099)	(0.0015)	(0.0019)
	Р	anel C. Finan	cial Risk Taking	ç
	Exit from the S	Stock Market	Wealth-to-In	come Ratio
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00258**	0.00322**	-0.01418**	-0.02485***
	(0.0011)	(0.0015)	(0.0059)	(0.0083)
Household Controls	Vos	Vos	Vos	Vos
Industry Controls	Vog	Vos	Vog	Vog
Clustoring	Muni	Muni	Muni	Muni
Maariaina litaa EE a	N	IVI UIII	No. 1	Mulli X-
Municipality FEs	res	res	res	res

Table O.A.16: Exposure to Robots and Household Economic Behavior - Sorting of Households

	IVEstimates	IVEstimates		
	(1)	(2)		
	Panel A. Labor Market			
	Change in Earnings	Transition into Unemployment		
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.01647**	0.00510***		
	(0.0069)	(0.0009)		
$\Delta EU_Import^{99 \rightarrow 07}$	-0.00111	-0.00234***		
	(0.0063)	(0.0008)		
	Panel B. Ho	usehold Wealth		
	Percentile Net Wealth Rank	Downward Mobility		
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.70335***	0.00449***		
	(0.0957)	(0.0015)		
$\Delta EU_Import^{99 \rightarrow 07}$	-0.52429***	-0.00539***		
	(0.1089)	(0.0017)		
	Panel C. Finar	ncial Risk Taking		
	Exit from the Stock Market	Change in Financial Wealth		
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00304**	-0.04844***		
	(0.0012)	(0.0099)		
$\Delta EU_Import^{99 \rightarrow 07}$	-0.00102	0.01008		
	(0.0011)	(0.0076)		
Household Controls	Yes	Yes		
Industry Controls	Yes	Yes		
Clustering	Muni	Muni		
Municipality FEs	Vos	Vos		
Municipanty PES	168	168		

Table O.A.17: Exposure to Robots and Household Economic Behavior - Controlling for Changes in Exposure to EU Import

	IVEstimates	s IV Estimates			
	(1)	(2)			
	Panel A. Labor Market				
	Change in Earnings	Transition into Unemployment			
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.01975*** 0.00532***				
	(0.0065)	(0.0009)			
	Panel B. Household Wealth				
	Percentile Net Wealth Rank	Downward Mobility			
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.93156***	0.00458***			
	(0.1024)	(0.0016)			
	Panel C. Financial Risk Taking				
	Exit from the Stock Market	Wealth-to-Income Ratio			
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00394*** -0.04238***				
	(0.0012)	(0.0066)			
	77				
Household Controls	Yes	Yes			
Industry Controls	Yes	Yes			
Clustering	Muni	Muni			
Municipality FEs	Yes Yes				

Table O.A.18: Exposure to Robots and Household Economic Behavior - Controlling for Unobserved Household Heterogeneity

	IVEstimates	IV Estimates IV Estimates			
	(1) (2)				
	Panel A. Labor Market				
	Change in Earnings	Transition into Unemployment			
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.01039 0.00278***				
	(0.0071)	(0.0008)			
	Panel B. Household Wealth				
	Percentile Net Wealth Rank	Downward Mobility			
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.60368***	0.00405**			
	(0.1105)	(0.0016)			
	Panel C. Financial Risk Taking				
	Exit from the Stock Market	Wealth-to-Income Ratio			
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00158	-0.03565***			
	(0.0012)	(0.0063)			
Household Controls	Vac	Vec			
Industry Controls	ies Vog	Ies Vez			
Industry Controls	les	ies			
Clustering	Muni Muni				
Municipality FEs	Yes Yes				

Table O.A.19: Exposure to Robots and Household Economic Behavior - Alternative Technological Frontier

	OLS Estimates	IVEstimates	OLS Estimates	IVEstimates	
	(1)	(2)	(3)	(4)	
	Panel A. Labor Market				
	Change in Earnings		Transition into Unemployment		
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.00859*	-0.01665**	0.00189***	0.00471***	
	(0.0046)	(0.0065)	(0.0006)	(0.0008)	
	Panel B. Household Wealth				
	Percentile Net Wealth Rank		Downward Mobility		
$\Delta Robot_Density^{99 \rightarrow 07}$	-0.44696***	-0.78879***	0.00252**	0.00362**	
	(0.0738)	(0.0978)	(0.0012)	(0.0016)	
	Panel C. Financial Risk Taking				
	Exit from the Stock Market		Wealth-to-Income Ratio		
$\Delta Robot_Density^{99 \rightarrow 07}$	0.00208**	0.00285***	-0.02126***	-0.03375***	
	(0.0009)	(0.0011)	(0.0044)	(0.0066)	
	\$7	37	37	\$7	
Household Controls	res	Yes	res	res	
Industry Controls	Yes	Yes	Yes	Yes	
Clustering	Muni	Muni	Muni	Muni	
Municipality FEs	Yes	Yes	Yes	Yes	

Table O.A.20: Exposure to Robots and Household Economic Behavior - Correcting the SEs at the Muni-Industry Level